ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Carbonell-Nicolau, Oriol

Working Paper Measuring Hierarchy

Provided in Cooperation with:

The Bichler & Nitzan Archives

Suggested Citation: Carbonell-Nicolau, Oriol (2024) : Measuring Hierarchy, The Bichler and Nitzan Archives, Toronto, https://oriolcn.github.io/papers/hierarchy.pdf, https://bnarchives.net/id/eprint/842/

This Version is available at: https://hdl.handle.net/10419/309442

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

NC ND http://creativecommons.org/licenses/by-nc-nd/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Measuring Hierarchy*

Oriol Carbonell-Nicolau[†]

15 October 2024

Abstract

This paper presents a novel axiomatic approach to measuring and comparing hierarchical structures. Hierarchies are fundamental across a range of disciplines—from ecology to organizational science—yet existing measures of hierarchical degree often lack systematic criteria for comparison. We introduce a mathematically rigorous framework based on a simple partial pre-order over hierarchies, denoted as \geq_H , and demonstrate its equivalence to intuitively appealing axioms for hierarchy comparisons.

Our analysis yields three key results. First, we establish that for fixed-size hierarchies, one hierarchy is strictly more hierarchical than another according to \geq_H if the latter can be derived from the former through a series of subordination removals. Second, we fully characterize the hierarchical pre-orders that align with \geq_H using two fundamental axioms: Anonymity and Subordination Removal. Finally, we extend our framework to varying-size hierarchies through the introduction of a Replication Principle, which enables consistent comparisons across different scales.

Keywords: hierarchy measurement, hierarchical pre-order, hierarchical index. *JEL classifications*: D23, L22.

1. Introduction

The concept of hierarchy is essential for understanding emergent properties in complex systems across a wide range of scientific disciplines. In ecology and earth sciences, hierarchical frameworks clarify the organization of ecosystems, encompassing everything from individual organisms to vast biomes. Similarly, social scientists utilize hierarchical models to examine power dynamics, social stratification, and organizational behavior within human societies. The evolution of economic entities—from simple partnerships to multinational corporations—has been marked by increasingly sophisticated hierarchical structures.

A critical examination of these hierarchical structures reveals both their advantages and disadvantages across various contexts. In biology, hierarchies can be beneficial by reducing conflict and enhancing group coordination among individuals. For instance, social hierarchies in animal species often lead to established dominance relationships that

^{*}This paper benefited from valuable comments and suggestions by the anonymous referees and the editors.

[†]Department of Economics, Rutgers University, 75 Hamilton St., New Brunswick, NJ 08901. E-mail: carbonell-nicolau@rutgers.edu.

minimize aggressive encounters and promote stability within groups. This stability can facilitate cooperative behaviors and improve resource allocation.

In organizational science and management, hierarchies establish clear lines of authority, accountability, and communication within institutions. This structured approach can enhance coordination, decision-making, and strategy implementation. However, such frameworks can also impede the pursuit of desirable goals. For instance, Wright (2024) argues that deeply entrenched knowledge-based hierarchies may yield negative epistemic consequences by reinforcing conservative selection biases against innovative research. This indicates that while hierarchies can streamline processes, they may also stifle creativity and adaptability.

Furthermore, it is possible that the evolution of hierarchical structures in human societies has significantly contributed to the historic development of income and wealth inequality, which has now reached unprecedented levels. Although authoritative evidence supporting this hypothesis is currently lacking, the notion aligns with some findings in the literature and remains intuitively compelling. This perspective raises critical questions about the social implications of hierarchical organization and its potential role in perpetuating disparities.

To explore the link between hierarchy and income distribution further, we note that hierarchical structures exhibit a strong correlation with earnings distributions. Some researchers have employed theoretical hierarchical models to explain observed worker compensation patterns. Empirical evidence suggests that the compensation of a firm's highest-paid official is primarily related to firm size, while other variables—particularly profit—have minimal explanatory power (Roberts, 1956). Simon (1957) proposes a simple hierarchical structure to elucidate the relationship between CEO compensation and firm size. Similarly, Lydall (1959) employs a comparable hierarchical mechanism to generate a labor income distribution whose upper tail aligns with empirical power law distributions.

More recently, Fix (2018, 2019) adapted the hierarchical models of Simon (1957) and Lydall (1959) in light of new data, finding that relative income within firms scales strongly with the average number of subordinates under an individual's control.

Both Simon (1957) and Lydall (1959) postulate a specific hierarchical structure in which each supervisor has the same number of immediate subordinates—a concept often referred to as the "span of control." Under this assumption, these authors express worker compensation as a function of both the span of control and the ratio of an individual's salary to those of their immediate subordinates. As noted by Simon (1957), the span of control serves as a measure of the "steepness" of organizational hierarchies. However, empirically, the span of control varies across ranks. This variability is documented in Fix (2018, 2019), where the span of control is replaced by the average number of subordinates.

Despite these insights, measures of hierarchical "steepness" remain "informal" due to the absence of a systematic criterion for comparing hierarchies. The distributional consequences of hierarchical structures are not well understood—both theoretically, owing to a lack of a general theory for hierarchy measurement, and empirically, due to the paucity of publicly available data.

Despite the pivotal role that hierarchies play in shaping organizational structures and influencing earnings distribution, our ability to systematically analyze and compare these structures remains limited. While several measures of hierarchical degree have been developed by researchers across various disciplines, they predominantly take the form of hierarchical indices that result in a complete ranking of hierarchical structures.¹ Although these measures may seem intuitively appealing, they often lack a solid theoretical foundation.

This paper represents a first attempt to advance beyond informal measures of hierarchical steepness by introducing a systematic, mathematically grounded approach. Methodolo-gically, our analysis is akin to the axiomatic underpinnings of inequality measurement (see, e.g., Cowell, 2016; Chakravarty, 2009, 2015).

The paper presents an axiomatic theory of hierarchy measurement, which serves as a crucial precondition for systematically analyzing how organizational design shapes economic outcomes and societal structures—particularly the relationship between income distribution and hierarchical frameworks. Furthermore, this work contributes to a broader body of literature, as hierarchies are fundamental to the evolution and transformation of societies as a whole. Indeed, the transition from small-scale to large-scale societies is closely linked with increasing hierarchical complexity and per capita energy capture (see, e.g., Turchin and Gavrilets, 2009; Fix, 2017; Graeber and Wengrow, 2021; Bichler and Nitzan, 2020).

Turning to the specifics of the formal analysis, we demonstrate that a simple partial pre-order defined over the set of hierarchies is equivalent to a set of intuitively appealing axioms or principles for hierarchy comparisons. This equivalence lays the groundwork for a comprehensive characterization of hierarchy measures that align with this partial pre-order.

We consider hierarchies represented as a series of nodes connected by paths. Each node symbolizes an individual, while the paths between nodes delineate the subordination relationships among them. Our focus is on hierarchies in which each subordinate has only one immediate supervisor.

A hierarchical pre-order \succeq is a reflexive and transitive binary relation over the set of all hierarchies. Because \succeq need not be complete, \succeq may render no judgment over some comparisons of hierarchies. When two hierarchies *h* and *h'* are comparable under \succeq , we write " $h \succeq h$ " to indicate that "*h* is at least as hierarchical as *h'*."

For hierarchies with the same number of individuals, we adopt two basic criteria for hierarchy comparisons. First, relabeling the individuals in a hierarchy does not alter its hierarchical structure. This property is called Anonymity. The second criterion is based on the notion of Subordination Removal. We say that a hierarchy h' is obtained from another hierarchy h by removal of a subordination relation if the sub-hierarchy h(i) of h that begins at an immediate subordinate i of a supervisor j in h is moved up one level in the hierarchy so that i is no longer an immediate subordinate of j, but rather either an unsupervised individual (if j has no supervisors) or an immediate subordinate of j's immediate supervisor. The sub-hierarchy h(i) remains otherwise intact, and the structure of h' is otherwise identical to that of h. The Subordination Removal postulate asserts that removing a subordination relation creates a less hierarchical structure.

We introduce a particular hierarchical pre-order, denoted by \succeq_H , which is instrumental in the formulation of our results. Specifically, we say that hierarchy h is at least as hierarchical as h' under \succeq_H (i.e., $h \succeq_H h'$) if there exists a bijection between the sets of individuals in both hierarchies with a particular property: for every individual i in h that corresponds to a subordinate j in h', the immediate supervisor of j in h' must correspond to

¹See, e.g., Mones et al. (2012); Corominas-Murtra et al. (2013); Krackhardt (1994); Trusina et al. (2004); Luo and Magee (2011); Czégel and Palla (2015).

a supervisor of i in h. This relationship not only establishes a one-to-one mapping between the individuals but also preserves the supervisory dynamics across the two hierarchies.

It turns out that the hierarchical pre-order \geq_H is closely related to the concept of subordination relation. Notably, a key result of this paper (Theorem 1) establishes that for any two hierarchies *h* and *h'* with the same number of individuals, *h* is strictly more hierarchical than *h'* under \geq_H (i.e., $h >_H h'$) if and only if *h'* can be derived from a relabeling of *h* through a series of removals of subordination relations.

A hierarchical pre-order is said to be \geq_H -consistent if it agrees with \geq_H for pairs of hierarchies that are comparable under \geq_H . The second primary result of this paper (Theorem 2) characterizes the class of \geq_H -consistent hierarchical pre-orders in terms of two axioms: for a fixed number of individuals, n, a hierarchical pre-order on the set of all hierarchies of size n satisfies Anonymity and Subordination Removal if and only if it is \geq_H -consistent.

Similar results can be obtained for extensions of hierarchical pre-orders to pairs of hierarchies of varying sizes. These results require the introduction of a third fundamental criterion alongside the existing Anonymity and Subordination Removal conditions: replicating a hierarchy—which yields two identical hierarchies, a superstructure that is itself a hierarchy in its own right—does not alter its hierarchical structure. This property is called Replication Principle.

Using the Replication Principle, the hierarchical pre-order \succeq_H can be extended to pairs of hierarchies of varying sizes as follows: $h \succeq_H h'$ if there exist two equally-sized replications, h_r and h'_r , of h and h', respectively, such that $h_r \succeq_H h'_r$.

The third main result of this paper (Theorem 3) states that a hierarchical pre-order on the set of all hierarchies (of any size) satisfies Anonymity, Subordination Removal, and the Replication Principle if and only if it is \succeq_H -consistent.

We examine two examples of \geq_H -consistent hierarchical pre-orders. The first compares hierarchies based on the number of supervisors between pairs of linked individuals. This pre-order represents a partial completion of \geq_H . The second example, which provides a complete ordering, is derived from a hierarchical index that computes the average number of supervisors for each hierarchy. Despite apparent similarities, our measure differs fundamentally from the hierarchical metric in Mones et al. (2012). We provide a thorough examination of this relationship within our framework in Section 4.

The literature on rank mobility (see, e.g., D'Agostino and Dardanoni, 2009; Bossert et al., 2016) is pertinent to the analysis presented in this paper.² Rank mobility examines how individuals transition across indicators of economic or social "status" over time. Since hierarchy inherently involves "rank" echelons that can be viewed as measures of "status," there is a potential link between measuring hierarchical degree and assessing rank mobility.

However, this connection is, at best, tenuous. In fact, a fundamental difference exists between rank mobility and the comparison of hierarchical structures, as illustrated by the following example.

Consider the three simple hierarchies depicted in Figure 1.

In hierarchies h and h'', there are two levels, or ranks. In hierarchy h, individual j is subordinate to individual i, while in h'', j supervises i. In contrast, hierarchy h' has no subordination relations, with both i and j occupying the same rank.

²For additional context, see the surveys on income mobility by Maasoumi (1998), Fields and Ok (1999), and Jäntti and Jenkins (2015).



Figure 1: Rank mobility vs. hierarchical measurement.

Notice that the transition from h to h'' involves both individuals moving across ranks, whereas only one individual changes rank in the transition from h to h'. This indicates that the shift from h to h'' reflects a greater degree of rank mobility. In fact, rank mobility is maximal in the transition from h to h''.

Despite the difference in rank mobility, our measures of hierarchical structure classify h and h'' as *equally* hierarchical. This is because both h and h'' share the same hierarchical structure and differ only in the labeling of individuals. According to the "Anonymity" axiom, this means that both hierarchies are essentially equivalent.

Thus, while h and h'' are maximally hierarchical and h' is minimally hierarchical, rank mobility is higher—indeed, maximal—in the transition from h to h''.

This analysis demonstrates that the degree of rank mobility between hierarchies is independent of their relative hierarchical structure. Specifically, we have shown that two equally hierarchical structures (h and h'') can be connected by a transition involving maximal rank mobility, while a transition to a less hierarchical structure (h') involves lower rank mobility.

The paper is structured as follows. Section 2 introduces the concept of hierarchy along with its relevant terminology. Section 3 formally defines (potentially incomplete) hierarchical pre-orders for hierarchies of fixed size. It presents the Anonymity and Subordination Removal axioms, and introduces a specific hierarchical pre-order, denoted by \geq_H , based on supervisory rank comparisons across hierarchies. This pre-order satisfies the aforementioned axioms (see Lemma 1) and is utilized in the first main result (Theorem 1) to fully characterize successive removals of subordination relations. The second main result (Theorem 2) demonstrates the equivalence between the Anonymity and Subordination Removal axioms and the \geq_H -consistency of a hierarchical pre-order—its alignment with \geq_H whenever two hierarchies can be compared under this pre-order. Section 3 concludes with a discussion on two specific (partial) completions of the hierarchical pre-order \geq_H .

Section 4 broadens the analysis from Section 3 to hierarchies of varying sizes. It introduces the Replication Principle and extends the hierarchical pre-order \geq_H to encompass hierarchies of any size, ensuring compliance with the Replication Principle, as well as the Anonymity and Subordination Removal axioms (see Lemma 2). The third main result (Theorem 3) establishes the equivalence of these three axioms with \geq_H -consistent hierarchical pre-orders. Additionally, Section 4 explores specific (partial) completions of \geq_H within the expanded domain of hierarchies of any size, comparing them to the hierarchical index proposed by Mones et al. (2012).

Finally, Section Section 5 summarizes our key findings and explores potential avenues for future research.

2. Hierarchies

We begin with the definition of a hierarchy.

Definition 1. A *hierarchy* is defined as a set of $n \in \mathbb{N}$ individuals satisfying the following conditions:

• There exists a set of level-0 individuals, I_0 , such that each $i \in I_0$ has either no subordinates or a set of level-1 subordinates, S_i , satisfying the following conditions:

-
$$S_i \cap I_0 = \emptyset$$
 for each $i \in I_0$.

- $S_i \cap S_{i'} = \emptyset$ for each *i*, *i'* ∈ I_0 with *i* ≠ *i'*.

In words, there is no overlap between the set of level-0 individuals, I_0 , and the set of i's subordinates, S_i ; and a subordinate of $i \in I_0$ in S_i cannot be, at the same time, a subordinate of another level-0 individual $i' \in I_0$.

- Suppose that the set of level-k individuals has been defined, where k ∈ {0, 1, ..., K−1}, and where K represents the total number of levels in the hierarchy. The set of level-(k + 1) individuals is defined as follows. Each level-(k + 1) subordinate, j, has either no subordinates or a set of level-(k + 2) subordinates, S_j, satisfying the following conditions:
 - For each level-(k + 1) subordinate j, the set of level-(k + 2) subordinates of j, S_j , contains no level- κ individuals, where $\kappa \in \{0, ..., k + 1\}$.
 - For any two distinct level-(k + 1) subordinates j and j', the sets of level-(k + 2) subordinates S_j and $S_{j'}$ of j and j', respectively, are disjoint.

For each level-*k* subordinate *i* in a given hierarchy *h*, where $k \in \{1, ..., K\}$ (and where *K* denotes the total number of levels in *h*), there is one level-(k - 1) supervisor, $p_h(i)$, one level-(k - 2) supervisor, $p_h^2(i)$, etc. The supervisor $p_h(i)$ of *i* is called *i*'s immediate supervisor. $p_h^2(i), p_h^3(i), ...$ are indirect supervisors of *i*, being supervisors of *i*'s immediate supervisor, $p_h(i)$.

In the sequel, the subscript "h" in expressions like $p_h(i)$ and $p_h^2(i)$ is sometimes omitted for simplicity. These omissions should not cause confusion, as the underlying hierarchy can be easily inferred from the context.

Individuals *i* and *j* in a given hierarchy are related if there is a path linking them, i.e., if either i = j or $j = p^{l}(i)$ for some *l*. If $i \neq j$, we say that *i* is a subordinate of *j*. If $i \neq j = p(i)$, we say that *i* is an immediate subordinate of *j*.

For each individual *i* in a hierarchy *h*, the *sub-hierarchy* containing *i* and all of *i*'s subordinates constitutes a properly defined hierarchy, denoted by h(i). The subordinates of *i* are the members of the sub-hierarchy h(i) other than *i*.

Note that, given a hierarchy *h* and its set of level-0 individuals, I_0 , *h* can be represented as a vector of sub-hierarchies $(h(i))_{i \in I_0}$. Note also that, given a hierarchy $h = (h(i))_{i \in I_0}$,



Figure 3: Relabeling.

 $(h(i))_{i \in I}$, where $I \subseteq I_0$, is a hierarchy in its own right, sometimes also referred to as a sub-hierarchy of h.

Hierarchies can be conveniently represented graphically as a series of nodes linked by paths. Each node represents an individual in the hierarchy. Figure 2 presents a hierarchy with two level-0 individuals, four level-1 individuals, and three level-2 individuals.

It should be noted that our definition of a hierarchy is not flexible enough to account for structures where a subordinate reports to multiple immediate supervisors.

3. Hierarchical pre-orders

To begin, we consider pre-orders on hierarchies of a fixed size. Extensions of these pre-orders to hierarchies of varying sizes are studied in Section 4.

Let \mathcal{H}_n be the set of all *n*-person hierarchies.

Definition 2. A *hierarchical pre-order* on \mathcal{H}_n is a reflexive and transitive binary relation \succeq on \mathcal{H}_n .

For $h, h' \in \mathscr{H}_n$, " $h \succeq h$ " means that "h is at least as hierarchical as h'."

Note that hierarchical pre-orders are not necessarily complete.

The symmetric and asymmetric parts of \succeq are denoted by \sim and \succ , respectively. For $h, h' \in \mathscr{H}_n$, the interpretation of the dominance relation " $h \succ h'$ " (resp., " $h \sim h'$ ") is that "h is more hierarchical than h'" (resp., "h and h' are equally hierarchical").

We now present two basic properties of hierarchical pre-orders.



Figure 4: Removing a subordination relation.

A hierarchy h' is said to be a *relabeling* of another hierarchy h if h' is obtained from h by relabeling the individuals in h. For example, the two hierarchies in Figure 3, h and h', are relabelings of each other.

Anonymity (A). A hierarchical pre-order \succeq on \mathcal{H}_n satisfies A if for any two hierarchies h and h' in \mathcal{H}_n , $h \sim h'$ whenever h' is a relabeling of h.

The Anonymity axiom asserts that, given a hierarchical pre-order, all relabelings of a given hierarchy belong to the same equivalence class. Note that relabelings do not alter the fundamental structure of a hierarchy; they simply rename the individuals involved.

We now introduce the notion of subordination relation removal.

Definition 3. We say that h' is obtained from a hierarchy h by *removing a subordination relation* if there exists a level-k subordinate i in h, where $k \in \{1, ..., K\}$ (with K denoting the total number of levels in the hierarchy h), satisfying the following conditions:

- If *i*'s immediate supervisor in *h*, *p_h(i)*, is a level-0 individual, then *h*' is the hierarchy in which the sub-hierarchy *h*(*i*) is no longer under *p_h(i)*'s supervision, *i* becomes a level-0 individual, and the sub-hierarchy that begins at *i* is *h*(*i*); *h*' is otherwise equal to *h*.
- If *i*'s immediate supervisor in *h*, *p_h(i)*, is a not level-0 individual, then *p_h(i)* is an immediate subordinate of *p_h²(i)*, i.e., *p_h(i)* ∈ *S_{p_h²(i)}*. In this case, *h*' is the hierarchy in which the sub-hierarchy *h(i)* is no longer under *p_h(i)*'s supervision, but rather under the direct supervision of *p_h²(i)*, so that *i* is no longer a level-*k* subordinate, but rather a level-(*k* − 1) subordinate in *S_{p_h²(i)}*, and the sub-hierarchy that begins at *i* is *h(i)*; *h*' is otherwise equal to *h*.

Figure 4 illustrates the previous definition. In Figure 4, h' can be obtained from h by taking the sub-hierarchy h(a) that starts at the node labeled "a" and moving it up so that the individuals in h(a) are no longer subordinates of the level-0 individual in h.

Figure 5 presents another example where individual *b* loses one subordinate, individual *c*, who becomes a direct subordinate of *b*'s supervisor, *a*.

The following axiom states that removing a subordination relation creates a less hierarchical structure.



Figure 5: Removing a subordination relation.

Subordination Removal (SR). A hierarchical pre-order \succeq on \mathcal{H}_n satisfies SR if for any two hierarchies *h* and *h'* in \mathcal{H}_n , h > h' whenever *h'* is obtained from *h* by removing a subordination relation.

We now define a particular hierarchical pre-order, denoted by \succeq_H , based on the comparison of supervisory ranks across hierarchies.

For any two hierarchies *h* and *h'* in \mathcal{H}_n , $h \succeq_H h'$ if and only if there exists a bijection ϕ from the set of individuals in *h* to the set of individuals in *h'* satisfying the following: for each individual *i* in *h* such that $\phi(i)$ is not a level-0 individual, the immediate supervisor of $\phi(i)$ in *h'*, $p_{h'}(\phi(i))$, is linked (via ϕ^{-1}) to a supervisor *j* of *i* in *h*, i.e., *j* is in the path from *i* to *i*'s level-0 supervisor: $\phi^{-1}(p_{h'}(\phi(i))) = j = p_h^l(i)$ for some *l*.

As an example, consider the two hierarchies given in Figure 6. The bijection ϕ from the set of individuals in *h* to the set of individuals in *h'* is represented by means of double-arrowed lines connecting nodes across the two hierarchies.

One can easily verify that the linked pairs across hierarchies depicted in Figure 6 satisfy the conditions from the definition of \geq_H . For example, take individual 2 in *h*, who is linked to individual *b* in *h'*, and whose immediate supervisor, *a*, is linked to individual 1 in *h*, who is a supervisor of 2. As another example, take individual 4 (in *h*), who is linked to individual *c* (in *h'*), whose immediate supervisor, *b*, is linked to 2 (in *h*), a supervisor of 4. Note that 2 lies in the path from 4 to 4's level-0 supervisor, 1, in *h*. Similar conditions can be verified for the other nodes in *h*. Thus, $h \geq_H h'$.

The symmetric and asymmetric parts of \succeq_H are denoted by \sim_H and \succ_H , respectively.

Our analysis reveals an intrinsic relationship between the hierarchical pre-order \succeq_H and our two basic axioms. In fact, \succeq_H satisfies both Anonymity and Subordination Removal.

Lemma 1. The hierarchical pre-order \succeq_H defined on \mathcal{H}_n is reflexive and transitive and satisfies A and SR.

The proof of Lemma 1 is relegated to Appendix A.2.

The notion of successive subordination removal, introduced next, plays an important role in our first main result.

For $h_1, h_L \in \mathcal{H}_n$, h_1 is obtained from h_L by *successive removals of subordination relations* if there are finitely many hierarchies $h_2, ..., h_{L-1}$ in \mathcal{H}_n such that h_l is obtained from h_{l+1} by removing a subordination relation, for each $l \in \{1, ..., L-1\}$.



Figure 7: Illustrating Theorem 1.

Our first main result provides a complete characterization of successive subordination relation removals in terms of the hierarchical pre-order \succeq_H .

Theorem 1. For $h, h' \in \mathcal{H}_n$, $h \succ_H h'$ if and only if h' can be obtained from some relabeling of h by successive removals of subordination relations.

The following discussion aims to offer visual and intuitive insights into Theorem 1. The formal proof is available in Appendix A.3.

Suppose that *h* and *h*' are hierarchies in \mathcal{H}_n . First, note that Theorem 1 is trivially true when n = 1 (i.e., when both *h* and *h*' are 1-person hierarchies). Proceeding by induction, we argue that Theorem 1 is true for arbitrary n > 1 if we know it is true for (n - 1)-sized hierarchies.

Consider the hierarchies *h* and *h'* depicted in Figure 7 and suppose that $h >_H h'$. Then $h \ge_H h'$, which implies that there exists a bijection ϕ mapping the individuals in *h* to those in *h'* with the following property: for every individual *i* in *h* such that $\phi(i)$ is not a level-0 individual in *h'*, the immediate supervisor of $\phi(i)$ in *h'* is linked (via ϕ^{-1}) to a supervisor of *i* in *h*.

Suppose that the bijection ϕ is represented by the double-arrowed lines connecting the nodes across the hierarchies, as shown in Figure 7. Now let $h \setminus \iota$ be the hierarchy obtained from *h* by removing individual ι , as shown in Figure 8. Note that all the direct subordinates of ι in *h* become level-0 individuals in $h \setminus \iota$.



Figure 8: Illustrating Theorem 1.

Similarly, let $h' \setminus \phi(\iota)$ denote the hierarchy formed by removing the individual $\phi(\iota)$ from h', as illustrated in Figure 8. This results in the two direct subordinates of $\phi(\iota)$ in h' becoming level-0 individuals, leading to a total of exactly four level-0 individuals in $h' \setminus \phi(\iota)$.

Note that the bijection represented in Figure 8 by the double-arrowed lines is the restriction of ϕ to the hierarchy $h \setminus \iota$. The properties of ϕ imply that for every individual *i* in $h \setminus \iota$ where $\phi(i)$ is not a level-0 individual in $h' \setminus \phi(\iota)$, the immediate supervisor of $\phi(i)$ in $h' \setminus \phi(\iota)$ is connected (through ϕ^{-1}) to a supervisor of *i* in $h \setminus \iota$.

Hence, $h \setminus \iota \succeq_H h' \setminus \phi(\iota)$. There are two cases to consider:

(i)
$$h \setminus \iota \succ_H h' \setminus \phi(\iota)$$
 and (ii) $h \setminus \iota \sim_H h' \setminus \phi(\iota)$.

In the first case, the induction hypothesis implies that $h' \setminus \phi(\iota)$ can be obtained from some relabeling of $h \setminus \iota$ by successive removals of subordination relations, since both $h \setminus \iota$ and $h' \setminus \phi(\iota)$ are (n - 1)-person hierarchies.

Note that, in the transition from $h \setminus \iota$ to $h' \setminus \phi(\iota)$, the removal of each subordination relation affects only the individuals within a single sub-hierarchy of $h \setminus \iota$. This means that the entire sequence of subordination removals from a relabeling of $h \setminus \iota$ to $h' \setminus \phi(\iota)$ can be decomposed into subsequences that transform relabeled sub-hierarchies of $h \setminus \iota$ into sub-hierarchies of $h' \setminus \phi(\iota)$.

For instance, in the specific case illustrated in Figure 8, this indicates that each of the four sub-hierarchies in $h' \setminus \phi(\iota)$ can be derived from a relabeled version of exactly one the four sub-hierarchies in $h \setminus \iota$ through a series of subordination relation removals.

Now, consider the hierarchy formed by adding the individual ι at the top of $h \setminus \iota$, as depicted in the left panel of Figure 9. Observe that the resulting hierarchy is precisely h. Similarly, let h'' be the hierarchy created by adding the individual $\phi(\iota)$ at the top of hierarchy $h' \setminus \phi(\iota)$, as shown in the right panel of Figure 9.

As we have noted, each of the four sub-hierarchies in $h' \setminus \phi(\iota)$ can be derived from a relabeling of exactly one of the four sub-hierarchies in $h \setminus \iota$ through a series of subordination relation removals. This implies that the hierarchy h'' from Figure 9 can be obtained from a relabeling of h by successive removals of subordination relations. This is not yet the conclusion we seek, as our goal is to demonstrate that h' (rather than h'') can be obtained from a relabeling of h through successive removals of subordination relations. However,



Figure 9: Illustrating Theorem 1.

note that in our example, h' can be derived from h'' by moving two of the four level-1 sub-hierarchies in h'' up to level 0.

Since h'' (resp., h') can be obtained from a relabeling of h (resp., h'') by successive removals of subordination relations, it follows that h' can also be obtained from a relabeling of h by successive removals of subordination relations, as we sought.

It remains to consider the case when $h \setminus \iota \sim_H h' \setminus \phi(\iota)$. To address this case, we rely on Lemma 4 (proven in Appendix A.1). In our context, this lemma states that $h \setminus \iota \sim_H h' \setminus \phi(\iota)$ implies that $h \setminus \iota$ is a relabeling of $h' \setminus \phi(\iota)$. Consequently, the hierarchies h and h'' depicted in Figure 9 are also relabelings of one another. Thus, since we have established that h' can be derived from h'' through successive removals of subordination relations, we conclude that h' can be obtained from a relabeling of h by successive removals of subordination relations, as we aimed to show.

Theorem 1 establishes the equivalence between two distinct concepts: the intuitive process of successively removing subordination relations within a hierarchy and the more formal notion embodied by the pre-order \succeq_H , which is defined through the comparison of supervisory ranks across different hierarchies.

A hierarchical pre-order \succeq on \mathcal{H}_n is \succeq_H -consistent if the following two conditions are satisfied for every pair h, h' in \mathcal{H}_n :

- $h \succ_H h' \Rightarrow h \succ h'$.
- $h \sim_H h' \Rightarrow h \sim h'$.

Thus, a \succeq_H -consistent hierarchical pre-order can only differ from \succeq_H for pairs of incomparable hierarchies under \succeq_H . In other words, \succeq_H -consistent hierarchical pre-orders are partial completions of \succeq_H .

The next main result states that the Anonymity and Subordination Removal axioms fully characterize \succeq_H -consistent hierarchical pre-orders.

Theorem 2. A hierarchical pre-order on \mathcal{H}_n satisfies A and SR if and only if it is \succeq_H -consistent.

The proof of Theorem 2, which builds on Theorem 1, is sufficiently straightforward to be included in the main text. This proof also references two lemmata proven in Appendix A.

Proof of Theorem 2. [*Sufficiency.*] Suppose that \succeq is \succeq_H -consistent. Because \succeq_H satisfies A and SR (Lemma 1), and since \succeq is \succeq_H -consistent, it follows that \succeq also satisfies A and SR.

[*Necessity*.] Suppose that \succeq is a hierarchical pre-order on \mathcal{H}_n satisfying A and SR. We must show that \succeq is \succeq_H -consistent, i.e., that the following two conditions are satisfied for every pair h, h' in \mathcal{H}_n :

- $h \succ_H h' \Rightarrow h \succ h'$.
- $h \sim_H h' \Rightarrow h \sim h'$.

Suppose first that $h \sim_H h'$. Then, by applying Lemma 4 (which is stated and proven in Appendix A.1), we can conclude that h is a relabeling of h'. Because \succeq satisfies A, it follows that $h \sim h'$.

Now suppose that $h >_H h'$. By Theorem 1, h' can be obtained from some relabeling of h, denoted by h^* , by successive removals of subordination relations. Therefore, there exist hierarchies h_1, \ldots, h_L in \mathcal{H}_n such that

$$h' \leftarrow_{RS} h_1 \leftarrow_{RS} \cdots \leftarrow_{RS} h_L \leftarrow_{RS} h^*,$$

where, for $\hat{h}, \overline{h} \in \mathcal{H}_n$, " $\hat{h} \leftarrow_{RS} \overline{h}$ " symbolically indicates that " \hat{h} can be obtained from \overline{h} by removing a subordination relation."

Consequently, because \succ satisfies SR, we have

$$h^* > h_L > \cdots > h_1 > h',$$

and since h^* is a relabeling of h and \succ satisfies A, we see that

$$h \sim h^* > h_L > \dots > h_1 > h'. \tag{1}$$

Because \succeq is reflexive and transitive, (1) implies that h > h' (Sen, 2017, Lemma 1*a, p. 56), as desired.

As an example of a \succeq_H -consistent hierarchical pre-order on \mathscr{H}_n , consider the pre-order \succeq_s defined as follows: $h \succeq_s h'$ if and only if there exists a bijection ϕ from the set of individuals in h to the set of individuals in h' such that, for each i in h, the number of supervisors of i in h is greater than or equal to the number of supervisors of $\phi(i)$ in h'.

In other words, *h* is considered more hierarchical than h' under \succeq_s if there exists a bijective correspondence between the individuals in the two hierarchies such that each individual *i* in *h* has at least as many (or more) supervisors compared to the individual in *h'* that corresponds to *i*.

The symmetric and asymmetric parts of \succeq_s are denoted, as usual, by \sim_s and \succ_s , respectively.

To illustrate, consider again the hierarchy from Figure 6. It is easy to see that the bijection represented via double-arrowed lines connecting nodes across hierarchies has the property that each linked pair of individuals has (weakly) more supervisors in *h* than in *h'*. Thus, $h \succeq_s h'$.

Like \succeq_H , the pre-order \succeq_s also satisfies Anonymity and Subordination Removal.

Proposition 1. *The hierarchical pre-order* \succeq_s *defined on* \mathcal{H}_n *is reflexive and transitive and satisfies* A *and* SR.

The proof of Proposition 1 is provided in Appendix A.4.



Figure 10: $h \succ_s h'$ and $h \not\geq_H h'$.

By Proposition 1 and Theorem 2, \succcurlyeq_s is \succcurlyeq_H -consistent, i.e., \succcurlyeq_s agrees with \succcurlyeq_H for pairs of hierarchies in \mathcal{H}_n that are comparable under \succcurlyeq_H .

The converse assertion is false, i.e., \succeq_H is not \succeq_s -consistent. In fact, in general, $h \succeq_s h'$ need not imply $h \succeq_H h'$. To see this, consider the two seven-person hierarchies depicted in Figure 10. It is easy to see that, for the bijection represented in the figure, each linked pair of individuals has (weakly) more supervisors in h than in h'. Thus, $h \succeq_s h'$.

However, $h \not\geq_H h'$. To understand this, we refer to Lemma 6 from Appendix A.1, which is restated here for the reader's convenience:

Suppose that I_0 (resp., I'_0) represents the set of level-0 individuals in h (resp., h'). Then $h \succeq_H h'$ if and only if there exists a finite partition of I'_0 consisting of $#I_0$ elements,

$$\{I'_1,\ldots,I'_{\#I_0}\},\$$

where $\#I_0$ denotes the cardinality of I_0 , such that for each $i \in I_0$, $h(i) \succeq_H (h'(\iota))_{\iota \in I'_i}$.

Note that, for the sub-hierarchy of h, \hat{h} , depicted in Figure 10, we cannot have $\hat{h} \succeq_H (h'(\iota))_{\iota \in I'}$ for $I' \subseteq I'_0$, since $\hat{h} \succeq_H (h'(\iota))_{\iota \in I'}$ implies that \hat{h} and $(h'(\iota))_{\iota \in I'}$ have the same size, and yet all the sub-hierarchies of h' with only one level-0 supervisor have more than two individuals.

Hence, $h \not\geq_H h'$, and Theorem 1 implies that h' cannot be obtained from some relabeling of *h* by successive removals of subordination relations.

Next, we consider complete orders over the set of hierarchies \mathcal{H}_n induced by hierarchical indices.

A *hierarchical index* on \mathscr{H}_n is a map $I : \mathscr{H}_n \to \mathbb{R}$ that assigns a "hierarchical degree" I(h) to every hierarchy $h \in \mathscr{H}_n$. The index I gives rise to a complete hierarchical order on \mathscr{H}_n , \succcurlyeq_I , defined as follows:

$$h \geq_I h' \Leftrightarrow I(h) \geq I(h').$$

Clearly, \succeq_I is a properly defined hierarchical order, i.e., a reflexive and transitive order on \mathcal{H}_n .



Figure 11: $h \succeq_{I_s} h'$ and $h \not\geq_s h'$.

For example, given $h \in \mathcal{H}_n$, let $s_h(i)$ represent the number of supervisors of i in h and define

$$I_s(h) = \frac{1}{n} \sum_{i \in h} s_h(i), \tag{2}$$

i.e., $I_s(h)$ denotes the average number of supervisors per individual in the hierarchy h.

Clearly, $\succeq_s \subseteq \succeq_{I_s}$, i.e., h is at least as hierarchical as h' under \succeq_{I_s} whenever h is at least as hierarchical as h' under \succeq_s . What is more, \succeq_{I_s} is \succeq_s -consistent. Indeed, for $h, h' \in \mathcal{H}_n$, $h \succ_s h'$ implies that there exists a bijection ϕ from the set of individuals in h to the set of individuals in h' such that

$$s_h(i) \ge s_{h'}(\phi(i)), \text{ for each } i \in h,$$

with the inequality being strict for at least one *i*. Consequently,

$$I_{s}(h) = \frac{1}{n} \sum_{i \in h} s_{h}(i) > \frac{1}{n} \sum_{i \in h} s_{h'}(\phi(i)) = \frac{1}{n} \sum_{i \in h'} s_{h'}(i) = I_{s}(h'),$$

whence $h >_{I_s} h'$. Similarly, once can show that $h \sim_s h'$ implies $h \sim_{I_s} h'$.

Conversely, \succcurlyeq_s is not \succcurlyeq_{I_s} -consistent. Indeed, \succcurlyeq_{I_s} -dominance need not imply \succcurlyeq_s -dominance, as the following example illustrates.

Consider the hierarchies h and h' depicted in Figure 11. It is easily verified that the average number of supervisors per individual in either h or h' is 3/5:

$$I_s(h) = I_s(h') = 3/5.$$

Hence, $h \sim_{I_s} h'$. Nevertheless, $h \not\sim_s h'$. Put differently, we have $h \succcurlyeq_{I_s} h'$ and yet $h \not\succcurlyeq_s h'$. To see that $h \not\succcurlyeq_s h'$, it suffices to note that there is one individual in h' who has two supervisors, while all individuals in h have at most one supervisor.

We have seen that \succeq_{I_s} is \succeq_s -consistent and that \succeq_{I_s} and \succeq_s are \succeq_H -consistent. Thus, \succeq_{I_s} , being a complete order on \mathcal{H}_n , is a completion of \succeq_H and \succeq_s .

Since \geq_{I_s} is reflexive, transitive, and \geq_H -consistent, Theorem 2 implies that \geq_{I_s} satisfies A and SR.



Figure 12: (h, h) is a replication of h.

Proposition 2. *The hierarchical order* \geq_{I_s} *defined on* \mathcal{H}_n *is reflexive and transitive and satisfies* A *and* SR.

4. Comparing hierarchies of varying sizes

In this section, we expand the previous analysis to encompass hierarchies of different sizes. Recall that \mathscr{H} is the set of *n* person biorarchies. The superset

Recall that \mathcal{H}_n is the set of *n*-person hierarchies. The superset

$$\mathscr{H} = \bigcup_n \mathscr{H}_n$$

represents the set of hierarchies of any size.

Definition 4. A hierarchical pre-order on \mathcal{H} is a reflexive and transitive binary relation on \mathcal{H} .

A *replication* of a hierarchy $h \in \mathcal{H}$ is a hierarchy in \mathcal{H} of the form (h, ..., h). By convention, h is a replication of itself.

For example, the ten-person hierarchy (h, h) in Figure 12 is a replication of the fiveperson hierarchy h.

The Replication Principle allows one to compare hierarchies in \mathcal{H} with their replications.

Replication Principle (RP). A hierarchical pre-order \succeq on \mathcal{H} satisfies **RP** if for any two hierarchies *h* and *h'* in \mathcal{H} , *h'* ~ *h* whenever *h'* is a replication of *h*.

This axiom asserts that, for a hierarchical pre-order on \mathcal{H} , any replication of a given hierarchy *h* is as hierarchical as *h*.

Note that, according to the Replication Principle, two hierarchies in \mathcal{H} can only be compared through replication if the number of individuals in one hierarchy is a multiple of the number in the other. For instance, if *h* is a two-person hierarchy, then any replication of *h* will result in a 2*n*-person hierarchy, where $n \in \mathbb{N}$.

The hierarchical pre-order \succeq_H on \mathcal{H}_n introduced in Section 3 can be extended to the domain \mathcal{H} as follows: for $h, h' \in \mathcal{H}, h' \succeq_H h$ if and only if there exists n such that h_r (resp., h'_r) is a replication of h (resp., h') in \mathcal{H}_n and $h'_r \succeq_H h_r$.

In other words: for $h, h' \in \mathcal{H}$, h' is at least as hierarchical as h if and only if there exist equally sized replications of h' and h, denoted h'_r and h_r respectively, such that h'_r is at least as hierarchical as h_r .

Lemma 1 implies that the extension of \succeq_H to \mathscr{H} is reflexive and transitive and satisfies A and SR.³ Moreover, the extension \succeq_H satisfies RP. Indeed, if h' = (h, ..., h) is a replication of h, then $h' \sim_H h$ because $(h, ..., h) \sim_H (h, ..., h)$.

Lemma 2. The hierarchical pre-order \succeq_H defined on \mathcal{H} is reflexive and transitive and satisfies A, SR, and RP.

Given the extension \succeq_H defined on \mathcal{H} , one can define \succeq_H -consistency as in the previous section.

A hierarchical pre-order \succeq on \mathcal{H} is \succeq_H -consistent if the following two conditions are satisfied for every pair h, h' in \mathcal{H} :

- $h \succ_H h' \Rightarrow h \succ h'$.
- $h \sim_H h' \Rightarrow h \sim h'$.

The following result states that the axioms A, SR, and RP fully characterize \succeq_{H^-} consistency for hierarchical pre-orders defined on \mathcal{H} . This result extends Theorem 2 to the domain \mathcal{H} .

Theorem 3. A hierarchical pre-order on \mathcal{H} satisfies A, SR, and RP if and only if it is \succeq_H -consistent.

Similar to Theorem 2, Theorem 3 can be proven using Theorem 1 and two ancillary lemmas proven in Appendix A.

Proof of Theorem 3. [*Sufficiency.*] Suppose that \succeq is \succeq_H -consistent. Because \succeq_H satisfies A, SR, and RP (Lemma 2), and since \succeq is \succeq_H -consistent, it follows that \succeq also satisfies A, SR, and RP.

[*Necessity*.] Suppose that \succeq is a hierarchical pre-order on \mathscr{H} satisfying A, SR, and RP. We must show that \succeq is \succeq_H -consistent, i.e., that the following two conditions are satisfied for every pair h, h' in \mathscr{H} :

- (a) $h \succ_H h' \Rightarrow h \succ h'$.
- (b) $h \sim_H h' \Rightarrow h \sim h'$.

³Indeed, given h, h', and h'' in \mathcal{H} , we have

$$\begin{array}{l} h \succcurlyeq_{H} h, \\ [h \succcurlyeq_{H} h' \& h' \succcurlyeq_{H} h''] \Rightarrow [h_{r} \succcurlyeq_{H} h'_{r} \& h'_{r} \succcurlyeq_{H} h''_{r}] \Rightarrow h_{r} \succcurlyeq_{H} h''_{r} \Rightarrow h \succcurlyeq_{H} h'' \\ \end{array}$$

where the relation " $h_r \succeq_H h_r$ " follows from the transitivity of \succeq_H restricted to the domain \mathcal{H}_n (Lemma 1).

To see that \succeq_H , defined on \mathscr{H} , satisfies A, suppose that h and h' are hierarchies in \mathscr{H} such that h' is a relabeling of h. Then h and h' have the same size, n. Since the restriction of \succeq_H to \mathscr{H}_n satisfies A (Lemma 1), we have $h \sim_H h'$.

To see that \succeq_H satisfies SR, suppose that h and h' are hierarchies in \mathcal{H} and that h' can be obtained from h by removing a subordination relation. Then h and h' have the same size, n. Since the restriction of \succeq_H to \mathcal{H}_n satisfies SR (Lemma 1), we have $h \succeq_H h'$.

Fix *h* and *h'* in \mathcal{H} . Suppose that $h \in \mathcal{H}_m$ and $h' \in \mathcal{H}_n$. Let h_r (resp., h'_r) be an *n*-times (resp., *m*-times) replication of *h* (resp., *h'*). Then h_r and h'_r are hierarchies in \mathcal{H}_{mn} .

Suppose first that $h \sim_H h'$. Since \succeq_H satisfies RP (Lemma 2),

$$h_r \sim_H h \sim_H h' \sim_H h'_r. \tag{3}$$

Because \succeq_H is reflexive and transitive (Lemma 2), \sim_H is transitive (Sen, 2017, Lemma 1*a, p. 56). Consequently, (3) implies that $h_r \sim_H h'_r$.

Since h_r , $h'_r \in \mathcal{H}_{mn}$ and $h_r \sim_H h'_r$, h_r is a relabeling of h'_r (see Lemma 4 in Appendix A.1). Because \succeq satisfies A and RP, it follows that

$$h \sim h_r \sim h'_r \sim h'$$
.

By transitivity of ~, we see that $h \sim h'$. This establishes (b).

Now suppose that $h \succ_H h'$. Let h_r (resp., h'_r) be an *n*-times (resp., *m*-times) replication of *h* (resp., *h'*). Since \succcurlyeq_H satisfies **RP** (Lemma 2),

$$h_r \sim_H h \succ_H h' \sim_H h'_r. \tag{4}$$

Because \geq_H is reflexive and transitive (Lemma 2), (4) gives $h_r >_H h'_r$ (Sen, 2017, Lemma 1*a, p. 56).

Since h_r , $h'_r \in \mathcal{H}_{mn}$ and $h_r >_H h'_r$, h'_r can be obtained from some relabeling of h_r , denoted by h^*_r , by successive removals of subordination relations (Theorem 1). Therefore, there exist hierarchies h_1, \ldots, h_L in \mathcal{H}_n such that

$$h'_r \leftarrow_{RS} h_1 \leftarrow_{RS} \cdots \leftarrow_{RS} h_L \leftarrow_{RS} h^*_r$$

where, for $\hat{h}, \overline{h} \in \mathcal{H}_{mn}, \ \hat{h} \leftarrow_{RS} \overline{h}$ " means that $\ \hat{h}$ can be obtained from \overline{h} by removing a subordination relation."

Consequently, because \succ satisfies SR,

$$h_r^* > h_L > \cdots > h_1 > h_r',$$

and since h_r^* is a relabeling of h_r and \succeq satisfies A, we see that

$$h_r \sim h_r^* > h_L > \dots > h_1 > h_r'. \tag{5}$$

Because \succeq is reflexive and transitive, (5) implies that $h_r > h'_r$ (Sen, 2017, Lemma 1*a, p. 56). Since \succeq satisfies **RP**, and since h_r (resp., h'_r) is a replication of h (resp., h'), it follows that

$$h \sim h_r > h'_r \sim h',$$

implying that h > h'. This establishes (a).

Recall the hierarchical pre-order \succeq_s on \mathscr{H}_n introduced in Section 3: $h \succeq_s h'$ if and only if there exists a bijection ϕ between the sets of individuals in the two hierarchies such that, for each *i* in *h*, the number of supervisors of *i* in *h* is greater than or equal to the number of supervisors of $\phi(i)$ in *h'*.

This pre-order can be extended to \mathcal{H} as follows: for $h, h' \in \mathcal{H}, h' \succeq_s h$ if and only if there exists n such that h_r (resp., h'_r) is a replication of h (resp., h') in \mathcal{H}_n and $h'_r \succeq_s h_r$.

Proposition 1 implies that the extension of \succeq_s to \mathscr{H} is reflexive and transitive and satisfies A and SR.⁴ In addition, the extension \succeq_s satisfies RP. Indeed, if h' = (h, ..., h) is a replication of h, then $h' \sim_s h$ because $(h, ..., h) \sim_s (h, ..., h)$.

Proposition 3. *The hierarchical pre-order* \succeq_s *defined on* \mathcal{H} *is reflexive and transitive and satisfies* A, SR, and RP.

By Proposition 3 and Theorem 3, \succeq_s is \succeq_H -consistent.

In Section 3, we demonstrated that for the hierarchical pre-orders \succeq_H and \succeq_s defined on \mathcal{H}_n , \succeq_s -dominance need not imply \succeq_H -dominance.⁵ This implies that the extensions \succeq_H and \succeq_s defined on \mathcal{H} share the same property. Consequently, since \succeq_s , defined on \mathcal{H} , is \succeq_H -consistent, \succeq_s aligns with \succeq_H for pairs of hierarchies that are comparable under \succeq_H , but there are cases where \succeq_H deems two hierarchies incomparable, whereas \succeq_s still orders them.

The hierarchical index I_s defined in (2) can also be extended to \mathcal{H} : for $h \in \mathcal{H}$, let

$$I_s(h) = \frac{1}{n} \sum_{i \in h} s_h(i), \tag{6}$$

where $s_h(i)$ denotes the number of supervisors of *i* in *h*. For each $h \in \mathcal{H}$, $I_s(h)$ represents the average number of supervisors per individual in the hierarchy *h*.

The hierarchical order \succeq_{I_s} induced by I_s on \mathcal{H} is defined by

$$h \succcurlyeq_{I_s} h' \Leftrightarrow I_s(h) \ge I_s(h').$$

The order \succeq_{I_s} agrees with \succeq_s for those pairs of hierarchies $h, h' \in \mathcal{H}$ that can be \succeq_s -ordered. In other words, \succeq_{I_s} is \succeq_s -consistent. To see this, suppose that $h \succ_s h'$ for $h, h' \in \mathcal{H}$. Then there exists n such that h_r (resp., h'_r) is a replication of h (resp., h') in \mathcal{H}_n and $h_r \succ_s h'_r$. Since $h_r, h'_r \in \mathcal{H}_n$ and $h_r \succ_s h'_r$, we have $I_s(h_r) > I_s(h'_r)$ (since we know, from the previous section, that the restriction of \succeq_{I_s} to \mathcal{H}_n is \succeq_s -consistent). Next, note that if $h \in \mathcal{H}_m$ and h_r is a k-times replication of h, so that mk = n, then

$$I_{s}(h_{r}) = \frac{1}{n} \sum_{i \in h_{r}} s_{h_{r}}(i) = \frac{1}{mk} k \sum_{i \in h} s_{h}(i) = I_{s}(h).$$

Similarly, $I_s(h'_r) = I_s(h')$. Consequently,

$$I_s(h) = I_s(h_r) > I_s(h'_r) = I_s(h'),$$

whence $h \succ_{I_s} h'$.

In a similar vein, one can show that $h \sim_s h'$, for $h, h' \in \mathcal{H}$, implies $h \sim_{I_s} h'$. Thus, \succeq_{I_s} is \succeq_s -consistent.

Conversely, \succeq_s is not \succeq_{I_s} -consistent. In fact, for $h, h' \in \mathcal{H}$, $h \succeq_{I_s} h'$ need not imply $h \succeq_s h'$. To illustrate this point, refer back to Figure 11 and its detailed explanation. Figure 11 represents two hierarchies h and h' in \mathcal{H} for which $h \sim_{I_s} h'$ and yet $h \neq_s h'$.

⁴This assertion can be proven using a method entirely analogous to that employed in the proof from Footnote 3, which demonstrates that the extension of \succeq_H to \mathscr{H} is reflexive, transitive, and satisfies the conditions A and SR.

⁵Refer to Figure 10 and its detailed explanation.

We have seen that \succeq_{I_s} is \succeq_s -consistent and that \succeq_{I_s} and \succeq_s are \succeq_H -consistent. Thus, \succeq_{I_s} , being a complete order on \mathcal{H} , is a completion of \succeq_H and \succeq_s .

Since \geq_{I_s} is reflexive, transitive, and \geq_H -consistent, Theorem 3 implies that \geq_{I_s} satisfies A, SR, and RP.

Proposition 4. *The hierarchical order* \geq_{I_s} *defined on* \mathcal{H} *is reflexive and transitive and satisfies* A*,* SR*, and* RP.

Remark 1. The hierarchical index $I_s : \mathcal{H} \to \mathbb{R}$ defined in (6) can be equivalently formulated in terms of the average number of subordinates. Formally, let $I_b : \mathcal{H} \to \mathbb{R}$ be defined by

$$I_b(h) = \frac{1}{n} \sum_{i \in h} b_h(i),$$

where $b_h(i)$ denotes the number of subordinates of *i* in *h*.

We claim that

$$I_s(h) = I_b(h), \text{ for every } h \in \mathcal{H}.$$
 (7)

We first establish the equivalence for hierarchies with exactly one level-0 individual. Consider the following induction argument. The equivalence is clearly true for any 1-person hierarchy. Now assume that (7) is satisfied for any (n - 1)-person hierarchy h with exactly one level-0 individual, where n > 1. We will show that (7) must also hold for any n-person hierarchy with exactly one level-0 individual.

Suppose that *h* is an *n*-person hierarchy, where n > 1, and that *h* has only one level-0 individual, ι . Let $h \setminus \iota$ denote the hierarchy derived from *h* by removing individual ι . Then $h \setminus \iota$ is an (n - 1)-person hierarchy and we have

$$I_{s}(h \setminus \iota) = \frac{1}{n-1} \sum_{i \in h \setminus \iota} s_{h \setminus \iota}(i) = \frac{1}{n-1} \sum_{i \in h \setminus \iota} b_{h \setminus \iota}(i) = I_{b}(h \setminus \iota)$$

by the induction hypothesis. Using this equality, we can write

$$\begin{split} I_{s}(h) &= \frac{1}{n} \sum_{i \in h} s_{h}(i) = \frac{1}{n} \sum_{i \in h \setminus \iota} (s_{h \setminus \iota}(i) + 1) = \frac{1}{n} \sum_{i \in h \setminus \iota} s_{h \setminus \iota}(i) + \frac{n - 1}{n} \\ &= \frac{n - 1}{n} \cdot I_{s}(h \setminus \iota) + \frac{n - 1}{n} = \frac{n - 1}{n} \cdot I_{b}(h \setminus \iota) + \frac{n - 1}{n} = \frac{1}{n} \sum_{i \in h \setminus \iota} b_{h \setminus \iota}(i) + \frac{n - 1}{n} \\ &= \frac{1}{n} \left(\sum_{i \in h \setminus \iota} b_{h \setminus \iota}(i) + n - 1 \right) = \frac{1}{n} \sum_{i \in h} b_{h}(i) = I_{b}(h), \end{split}$$

as desired.

Now let *h* be an arbitrary *n*-person hierarchy. Then *h* can be expressed as

$$h=(h(i))_{i\in I_0},$$

where I_0 denotes the set of all level-0 individuals in h. Let $n_{h(i)}$ represent the number of individuals in h(i) for $i \in I_0$. Because

$$I_s(h(i)) = I_b(h(i)), \text{ for all } i \in I_0,$$

we have

$$\begin{split} I_{s}(h) &= \frac{1}{n} \sum_{j \in h} s_{h}(j) = \frac{1}{n} \sum_{i \in I_{0}} \sum_{j \in h(i)} s_{h}(j) = \frac{1}{n} \sum_{i \in I_{0}} [n_{h(i)}I_{s}(h(i))] = \frac{1}{n} \sum_{i \in I_{0}} [n_{h(i)}I_{b}(h(i))] \\ &= \frac{1}{n} \sum_{i \in I_{0}} \sum_{j \in h(i)} b_{h}(j) = \frac{1}{n} \sum_{j \in h} b_{h}(j) = I_{b}(h), \end{split}$$

which establishes the desired conclusion.

The hierarchical index $I_s : \mathscr{H} \to \mathbb{R}$ defined in (6) bears certain similarities to the "global reaching centrality" measure introduced by Mones et al. (2012). This measure is defined for unweighted directed graphs—a framework that naturally encompasses our notion of hierarchies. Indeed, any hierarchy can be mapped to a directed graph by replacing the paths linking individuals with directed edges (one-sided arrows). This mapping can be constructed in two distinct ways: either by directing arrows from each supervisor to their immediate subordinates, or conversely, from each subordinate to their immediate supervisor.

Using the subordinate-to-supervisor mapping, we can express the core concept from Mones et al. (2012) as follows. For an individual *i* in an *n*-person hierarchy *h*, the "local reaching centrality" $C_h(i)$ is defined as the proportion of all individuals in *h*, other than *i*, that can be reached from *i* via outgoing edges. More formally, $C_h(i)$ is calculated by dividing the number of individuals reachable from *i* through outgoing edges by the total number of individuals in *h* minus 1 (excluding *i* itself from both the numerator and denominator).

Let C_h denote the maximum local reaching centrality among all individuals in hierarchy h. The global reaching centrality of h, denoted by GRC(h), is then defined as the average deviation of individuals' local reaching centrality from this maximum:

$$GRC(h) = \frac{\sum_{i=1}^{n} (\overline{C}_{h} - C_{h}(i))}{n-1},$$

where *n* is the number of individuals in the hierarchy.

To illustrate this concept, consider the five-person hierarchy h depicted in Figure 13. In this representation, directed edges indicate subordination relations, with arrows pointing from subordinates to their immediate supervisors. This example allows us to demonstrate the calculation of both local reaching centralities for individual nodes and the global reaching centrality for the entire hierarchy.

In Figure 13, local reaching centralities are displayed next to each individual's node. As can be observed, the maximum local reaching centrality \overline{C}_h is 1/2. Using these values, we can compute the global reaching centrality:

$$GRC(h) = \frac{\frac{1}{2} - \frac{1}{2}}{4} + \frac{\frac{1}{2} - \frac{1}{4}}{4} + \frac{\frac{1}{2} - 0}{4} + \frac{\frac{1}{2} - 0}{4} + \frac{\frac{1}{2} - 0}{4} = \frac{7}{16}.$$

We note that

$$C_h(i) = \frac{s_h(i)}{n-1}$$
, for each individual *i* in *h*,

....



Figure 13: Calculating *GRC*_h.

where $s_h(i)$ represents the number of supervisors of *i* in *h*. With this foundation, we can now explore the relationship between the individual supervision index $I_s(h)$ and the group relative centrality index GRC(h) as follows:

$$GRC(h) = \frac{\sum_{i=1}^{n} (\overline{C}_{h} - C_{h}(i))}{n-1} = \frac{n\overline{C}_{h} - \sum_{i=1}^{n} C_{h}(i)}{n-1} = \frac{n\overline{C}_{h} - \frac{1}{n-1} \sum_{i=1}^{n} s_{h}(i)}{n-1} = \frac{n\overline{C}_{h} - \frac{1}{n-1} \sum_{i=1}^{n} s_{h}(i)}{n-1} = \frac{n\overline{C}_{h} - \frac{n}{n-1} I_{s}(h)}{n-1} = \frac{n}{n-1} \left(\overline{C}_{h} - \frac{I_{s}(h)}{n-1}\right).$$
(8)

The relationship between $I_s(h)$ and GRC(h) described by (8) implies that for pairs of equally-sized hierarchies h and h' in \mathcal{H} with identical maximum local reaching centralities, i.e., satisfying $\overline{C}_h = \overline{C}_{h'}$, we have

$$GRC(h) \ge GRC(h') \Leftrightarrow I_s(h') \ge I_s(h).$$
 (9)

However, this equivalence no longer holds for pairs of hierarchies with different maximum local reaching centralities. In fact, the global reaching centrality index fails to be \succcurlyeq_H -consistent.

To see this, consider first the two three-person hierarchies represented in Figure 14. These hierarchies have the same number of individuals and the same maximum local reaching centralities, since

$$\overline{C}_h = \overline{C}_{h'} = \frac{1}{2}.$$

Consequently, (9) holds, and so

$$I_s(h') = \frac{2}{3} > \frac{1}{3} = I_s(h')$$

implies that GRC(h) < GRC(h').

On the other hand, it is easy to see that $h \succeq_H h'$ (for example, using the bijection depicted in Figure 14) and yet $h' \not\geq_H h$, so that $h \succ_H h'$. To see that $h' \not\geq_H h$, consider that any bijection between the set of individuals in h' and those in h must associate a level-0



Figure 14: $h \succ_H h'$ and GRC(h) < GRC(h').



Figure 15: $h >_H h'$ and GRC(h) > GRC(h').

individual in h', denoted as i, with a level-1 individual in h. However, the supervisor of this level-1 individual cannot be linked to i or any supervisor of i.

Thus, we have $h \succ_H h'$ and GRC(h) < GRC(h').

Now consider the two hierarchies, h and h', depicted in Figure 15. We can establish that $h \succeq_H h'$ using the bijection illustrated in the figure. However, the reverse relation does not hold: $h' \not\geq_H h$. To see why, consider any bijection from the individuals in h' to those in h. Such a bijection must map some level-1 individual i in h' to a subordinate in h. This subordinate in h necessarily has a supervisor that does not correspond to any supervisor of i in h', thus violating the conditions for $h' \succeq_H h$.

Note that the maximum local reaching centralities differ across hierarchies in this example:

$$\overline{C}_{h'} = \frac{1}{2} < 1 = \overline{C}_h.$$
$$I_s(h') = \frac{2}{3} < 1 = I_s(h).$$

Since

we can use (8) to compute the corresponding global reaching centralities:

$$GRC(h') = \frac{1}{4} < \frac{3}{4} = GRC(h).$$

This example, in conjunction with the previous one, yields a key insight into the relationship between hierarchical structures and centrality measures. Firstly, we observe that $h >_H h'$, indicating that hierarchy h is strictly more hierarchical than h' according to \succcurlyeq_H . Secondly, we find that the global reaching centrality measures of these hierarchies satisfy GRC(h') < GRC(h). This ordering stands in contrast to our previous example, where the relationship between global reaching centrality and \succcurlyeq_H was reversed. This reversal demonstrates that the global reaching centrality index is not \succcurlyeq_H -consistent. In other words, a more hierarchical structure as defined by \succcurlyeq_H does not necessarily imply a higher global reaching centrality.

Alternatively, we can reformulate global reaching centrality by considering the downward flow of authority rather than the upward chain of command, i.e., using directed edges that flow from supervisors to their immediate subordinates, without altering the key insights derived from our previous analysis. This alternative formulation provides a different perspective on hierarchical relationships while maintaining consistency with our established findings regarding the non-monotonicity of global reaching centrality with respect to \succeq_H .

Indeed, this alternative formulation of global reaching centrality yields consistent results with our previous findings. Examining the example in Figure 14, we observe that

$$h \succ_H h'$$
 and $GRC(h) > GRC(h')$,

indicating that the more hierarchical structure h corresponds to a higher global reaching centrality. Conversely, the example from Figure 15 demonstrates that

$$h \succ_H h'$$
 and $GRC(h') \geq GRC(h)$.

Here, despite *h* being more hierarchical, it exhibits a lower global reaching centrality than *h*'. This juxtaposition of results clearly illustrates the non-monotonic relationship between hierarchical structure and global reaching centrality, reinforcing our earlier conclusion that *GRC* is not \succeq_H -consistent, even under this alternative formulation.

5. Concluding remarks

This paper introduces a novel axiomatic framework for measuring and comparing hierarchical structures. By establishing a mathematical foundation for hierarchy comparison, our work represents a first step towards a systematic analysis of how organizational architecture influences economic outcomes and shapes societal structures.

At the core of our framework are three novel axiomatic principles that provide a basis for systematic hierarchy comparison. The Anonymity axiom establishes that the underlying hierarchical structure remains invariant under any relabeling of individuals within the hierarchy, ensuring that the hierarchical nature of an organization is determined by its structural relationships rather than by the specific identities of its members. The Subordination Removal axiom formalizes the intuitive notion that eliminating a supervisory relationship necessarily results in a less hierarchical structure, providing a crucial basis for comparing the relative "steepness" of different hierarchies. Finally, the Replication Principle extends our framework beyond the constraints of fixed-size comparisons by stipulating that replicating a hierarchy—creating multiple identical copies of its structure—preserves its hierarchical degree, thus enabling meaningful comparisons between hierarchies of different sizes while maintaining consistency with our other axioms.

We introduce a new and straightforward hierarchical pre-order, \geq_H , defined through the comparison of supervisory ranks across hierarchies. We demonstrate that this pre-order is intrinsically related to our axioms. Specifically, we show that one hierarchy is strictly more hierarchical than another under \geq_H if and only if the latter can be derived from the former—up to relabeling—through a series of successive subordination removals.

We define \geq_H -consistent hierarchical pre-orders as those that align with \geq_H when comparing two hierarchies that can be ranked under \geq_H . We characterize all \geq_H -consistent hierarchical pre-orders via the Anonymity and Subordination Removal axioms and extend our framework to accommodate hierarchies of varying sizes through the Replication Principle.

Our analysis reveals that the notion of hierarchical degree encapsulated in our basic axioms is fundamentally distinct from the essence of rank mobility measurement. Additionally, we study examples of partial and full completions of the core hierarchical pre-order \succcurlyeq_H , comparing them to existing hierarchical indices.

The theoretical foundations established in this work naturally point to a number of avenues for future research. The first involves expanding our framework to encompass networks with multiple immediate supervisors, as the current focus on hierarchies with single immediate supervisors may not capture all real-world organizational structures.

The second avenue leverages our framework's refined hierarchical comparison methods to explore the interplay between organizational design and socioeconomic outcomes. This research direction examines the broad implications of organizational structure for income distribution and social dynamics.

Of particular interest is the prospect of a nuanced theoretical analysis investigating the correlation between "deeper hierarchy" and established measures of income inequality, poverty, and polarization. Even in the context of "economic growth," deeply entrenched hierarchies can hinder equitable resource distribution. In such scenarios, similar levels of absolute poverty become increasingly normatively regrettable. Understanding these determinants requires a comprehensive analysis of how power dynamics and social stratification might counteract poverty alleviation efforts, drawing on existing research about poverty reduction failure.⁶

A. Appendix

A.1. Preliminary lemmata

Lemma 3. For $h, h' \in \mathscr{H}_n$, $h \sim_H h'$ implies that there exists a bijection ϕ from the set of individuals in h to the set of individuals in h' satisfying the following:

(a) For each level-k individual i in h, $\phi(i)$ is a level-k individual in h'.

⁶Poverty reduction failure is studied in Kanbur and Mukherjee (2007) and Chakravarty and D'Ambrosio (2013).

- (b) For each individual *i* in *h*, the number of immediate subordinates of *i* in *h* equals the number of immediate subordinates of $\phi(i)$ in *h'*.
- (c) For each individual i in h, the set

$$\phi(I_{h(i)}) = \{\phi(i) : i \in I_{h(i)}\},\$$

where $I_{h(i)}$ denotes the set of all individuals in the sub-hierarchy h(i), is equal to the set of all individuals in the sub-hierarchy $h'(\phi(i))$.

Proof. Since $h \succeq_H h'$, there exists, by definition, a bijection ϕ from the set of individuals in h to the set of individuals in h' satisfying the following:

(I) For each individual *i* in *h* such that $\phi(i)$ is not a level-0 individual, the immediate supervisor of $\phi(i)$ in *h'*, $p_{h'}(\phi(i))$, links (via ϕ^{-1}) to a supervisor *j* of *i* in *h*: $\phi^{-1}(p_{h'}(\phi(i))) = j = p_h^l(i)$ for some *l*.

Similarly, since $h' \succeq_H h$, there exists a bijection ϕ' from the set of individuals in h' to the set of individuals in h satisfying the following: for each individual i in h' such that $\phi'(i)$ is not a level-0 individual, the immediate supervisor of $\phi'(i)$ in h, $p_h(\phi'(i))$, links (via ϕ'^{-1}) to a supervisor j of i in h': $\phi'^{-1}(p_h(\phi'(i))) = j = p_{h'}^l(i)$ for some l.

Let I_0 (resp., I'_0) be the set of all level-0 individuals in h (resp., h').

First, we show that

$$\phi(I_0) = \{\phi(i) : i \in I_0\} \subseteq I'_0$$

To see this, note that $j \in \phi(I_0) \setminus I'_0$ implies that there exist $i \in I_0$ and a level-*k* individual *j* in *h'*, where k > 0, such that $\phi(i) = j$. But then $\phi^{-1}(p_{h'}(\phi(i))) \neq p_h^l(i)$ for any *l*, which contradicts (I). Therefore, $\phi(I_0) \setminus I'_0 = \emptyset$, which implies that $\phi(I_0) \subseteq I'_0$.

Similarly, we can show that $\phi'(I'_0) \subseteq I_0$.

Next, let I_l (resp., I'_l) be the set of all level-l individuals in h (resp., h'). Suppose that the containments $\phi(I_l) \subseteq I'_l$ and $\phi'(I'_l) \subseteq I_l$ have been proven for each $l \in \{0, ..., k\}$ and some $k \ge 0$. Then the containments $\phi(I_{k+1}) \subseteq I'_{k+1}$ and $\phi'(I'_{k+1}) \subseteq I_{k+1}$ can also be proven.

To see this, note first that, for each l, the two containments $\phi(I_l) \subseteq I'_l$ and $\phi'(I'_l) \subseteq I_l$ imply that h and h' have the same number of level-l individuals. Indeed, if there were more level-l individuals in h', then $\phi(I_l)$ would be a strict subset of I'_l , and, since both I_l and $\phi(I_l)$ have the same cardinality, I_l would be a smaller set than I'_l , contradicting the containment $\phi'(I'_l) \subseteq I_l$. A similar contradiction can be obtained under the assumption that there are more level-l individuals in h.

Now, if $j \in \phi(I_{k+1}) \setminus I'_{k+1'}$, since $\phi(I_l) \subseteq I'_l$ and I_l and I'_l have the same cardinality for each $l \in \{0, ..., k\}$, we see that $j \in \phi(I_{k+1}) \setminus (\bigcup_{l=0}^{k+1} I'_l)$. Consequently, there exist $i \in I_{k+1}$ and a level- κ' individual j in h', where $\kappa' > k + 1$, such that $\phi(i) = j$. But then $\phi^{-1}(p_{h'}(\phi(i)))$ must be a level- κ individual in h, where $\kappa \ge k + 1$. Indeed, if $\phi^{-1}(p_{h'}(\phi(i)))$ where a level-l individual in h for some $l \in \{0, ..., k\}$, then $\phi(I_l) \nsubseteq I'_l$ (since $p_{h'}(\phi(i)) = p_{h'}(j)$ is a level- $(\kappa' - 1)$ individual in h' and $\kappa' > k + 1$), contradicting the assumed containment $\phi(I_l) \subseteq I'_l$. Consequently, $\phi^{-1}(p_{h'}(\phi(i))) \neq p_h^l(i)$ for any ℓ , which contradicts (I). Therefore, $\phi(I_{k+1}) \setminus I'_{k+1} = \emptyset$, which implies that $\phi(I_{k+1}) \subseteq I'_{k+1}$.

Next, fix a level-*k* individual *i* in *h*. Since $\phi(I_k) \subseteq I'_k$, it follows that $\phi(i)$ is a level-*k* individual in *h'*. This establishes (a).

To see that (b) holds, let *i* be an individual in *h*. Suppose that *i* is a level-*k* individual. Proceeding by contradiction, suppose that the number of immediate subordinates of *i* in *h* is not equal to the number of immediate subordinates of $\phi(i)$ in *h'*. If he number of immediate subordinates of $\phi(i)$ is greater, then (by (a)) there exists a subordinate *j* of $\phi(i)$ linking (via ϕ^{-1}) to a level-(*k* + 1) subordinate *i* in *h* whose immediate supervisor, *i*^{*}, is not *i*. But then $\phi(\iota) = j$ and $\phi(i)$ is *j*'s immediate supervisor in *h'*, and yet $\phi(i)$ links (via ϕ^{-1}) to $i \neq i^*$, implying that *i* is not a supervisor of *ι*, which contradicts (I).

Hence, the number of immediate subordinates of *i* in *h* must be greater than or equal to the number of immediate subordinates of $\phi(i)$ in *h*'.

If the number of immediate subordinates of *i* in *h* is greater than the number of immediate subordinates of $\phi(i)$ in *h'*, there exists an immediate subordinate *ι* of *i* in *h* such that $\phi(\iota)$'s immediate supervisor in *h'*, $p_{h'}(\phi(\iota))$, is not $\phi(i)$. But then $\phi^{-1}(p_{h'}(\phi(\iota)))$ is a level-*k* individual different from *i*, implying that $\phi^{-1}(p_{h'}(\phi(\iota)))$ is not a supervisor of *ι* in *h*, which contradicts (I).

Thus, the number of immediate subordinates of *i* in *h* is equal to the number of immediate subordinates of $\phi(i)$ in *h'*. This establishes (b).

It only remains to prove (c). Fix an individual *i* in *h*, and let $I_{h(i)}$ (resp., $I_{h'(\phi(i))}$) be the set of all individuals in the hierarchy h(i) (resp., $h'(\phi(i))$). We must show that $\phi(I_{h(i)}) = I_{h'(\phi(i))}$.

Suppose that *i* is a level-*k* individual. Note that it suffices to prove the following: Suppose that *j* is a level-(*k* + *l*) individual in *h*(*i*) for $l \ge 0$. Then $\phi(S_j) = S_{\phi(j)}$, where S_j (resp., $S_{\phi(j)}$) represents the set of immediate subordinates of *j* (resp., $\phi(j)$) in *h* (resp., *h'*).

Suppose that *j* is a level-(*k* + *l*) individual in *h*(*i*) for $l \ge 0$. Suppose that there exists $\iota \in S_j$ such that $\phi(\iota) \notin S_{\phi(j)}$. Then, since ι is a level-(*k* + *l* + 1) individual in *h*, so that $\phi(\iota)$ is a level-(*k* + *l* + 1) individual in *h'* (by (a)), $\phi(\iota)$'s immediate supervisor in *h'*, $p_{h'}(\phi(\iota))$, is a level-(*k* + *l*) individual in *h'* who links (via ϕ^{-1}) to a level-(*k* + *l*) individual in *h*, $\phi^{-1}(p_{h'}(\phi(\iota)))$. Note that $\phi^{-1}(p_{h'}(\phi(\iota)))$, being different from *j* (since $\phi(\iota) \notin S_{\phi(j)}$ and so $p_{h'}(\phi(\iota)) \neq \phi(j)$), is not a supervisor of ι in *h*. Since this contradicts (I), we see that $\phi(S_j) \subseteq S_{\phi(j)}$. But then $\phi(S_j) = S_{\phi(j)}$, since S_j and $S_{\phi(j)}$ (and hence $\phi(S_j)$ and $S_{\phi(j)}$) have the same cardinality (by (b)).

Lemma 4. For $h, h' \in \mathcal{H}_n$, $h \sim_H h'$ implies that h is a relabeling of h'.

Proof. It suffices to show that there exists a bijection ϕ from the set of individuals in h to the set of individuals in h' such that h(i) is a relabeling of $h'(\phi(i))$ for each i in h.

Let ϕ be the bijection given by Lemma 3. Let *K* be the largest level for which there are level-*K* individuals in *h*. Then all the level-*K* individuals in *h* have zero subordinates. By item (a) of Lemma 3, for any level-*K* individual *i* in *h*, $\phi(i)$ is a level-*K* individual in *h*'; moreover, since $\phi(i)$ has zero subordinates, item (b) of Lemma 3 implies that h(i) is a relabeling of $h'(\phi(i))$.

Suppose that h(i) has been shown to be a relabeling of $h'(\phi(i))$ for each level-k individuals i in h, where $k \in \{K, K-1, ..., 1\}$. Then h(i) is a relabeling of $h'(\phi(i))$ for each level-(k - 1) individual i in h.

To see this, fix a level-(k - 1) individual *i* in *h*. Let S_i (resp., S'_i) be the set of level-*k* subordinates of *i* (resp., $\phi(i)$) in *h* (resp., *h'*). If

$$\phi(S_i) = \{\phi(j) : j \in S_i\} = S'_i$$

were true, then, because h(j) is a relabeling of $h'(\phi(j))$ for each $j \in S_i$, it would follow that h(i) is a relabeling of $h'(\phi(i))$. Thus, it suffices to show that $\phi(S_i) = S'_i$.

By items (a) and (c) of Lemma 3, we know that $\phi(S_i)$ is a set of level-*k* individuals in *h*' contained in the set of all individuals in the sub-hierarchy $h'(\phi(i))$. Since S'_i is the set of all level-*k* individuals in $h'(\phi(i))$, item (a) of Lemma 3 gives $\phi(S_i) \subseteq S'_i$. But then $\phi(S_i) = S'_i$, since S_i and S'_i (and hence $\phi(S_i)$ and S'_i) have the same cardinality (by item (b) of Lemma 3).

Lemma 5. For $h, h' \in \mathcal{H}_n$, if h' can be obtained from h by removing a subordination relation, then $h >_H h'$.

Proof. We proceed by induction on *n*. The statement is clearly true if n = 1. We now prove the statement for any n > 1 under the assumption that it is true for *m*-person hierarchies, where $m \in \{1, ..., n - 1\}$.

Because h' can be obtained from h by removing a subordination relation, there exists a level-k subordinate i^* in h, where $k \in \{1, ..., K\}$ (and where K denotes the total number of levels in the hierarchy h), satisfying the following:

- (i) If i^* 's immediate supervisor, $p_h(i^*)$, is a level-0 individual in h, then h' is the hierarchy in which the sub-hierarchy $h(i^*)$ is no longer under $p_h(i^*)$'s supervision, i^* becomes a level-0 individual, and the sub-hierarchy that begins at i^* is $h(i^*)$; h' is otherwise equal to h.
- (ii) If *i**'s immediate supervisor in *h*, $p_h(i^*)$, is a not level-0 individual, then $p_h(i^*)$ is an immediate subordinate of $p_h^2(i^*)$, i.e., $p_h(i^*) \in S_{p_h^2(i^*)}$. In this case, *h*' is the hierarchy in which the sub-hierarchy $h(i^*)$ is no longer under $p_h(i^*)$'s supervision, but rather under the direct supervision of $p_h^2(i^*)$, so that *i** is no longer a level-*k* subordinate, but rather a level-(k 1) subordinate in $S_{p_h^2(i^*)}$, and the sub-hierarchy that begins at *i** is $h(i^*)$; *h*' is otherwise equal to *h*.

First, we show that $h \succeq_H h'$. To see this, let ϕ be the identity map from the set of individuals in *h* to the set of individuals in *h'*. It suffices to prove the following:

(*) For each individual *i* in *h* such that $\phi(i)$ is not a level-0 individual, the immediate supervisor of $\phi(i)$ in *h*', $p_{h'}(\phi(i))$, links (via ϕ^{-1}) to a supervisor *j* of *i* in *h*: $\phi^{-1}(p_{h'}(\phi(i))) = j = p_h^l(i)$ for some *l*.

Note that if the sub-hierarchy $h(i^*)$ is removed from h and the sub-hierarchy $h'(\phi(i^*)) = h'(i^*)$ is removed from h', the remaining hierarchies are identical. Therefore, for any individual i in h not in $h(i^*)$, (*) holds.

Next, fix an individual *i* in $h(i^*)$. If $i \neq i^*$, then, since the two sub-hierarchies $h(i^*)$ and $h'(i^*)$ are identical, and since ϕ is the identity map, (*) holds.

It remains to prove (*) for $i = i^*$. Note that if $\phi(i^*) = i^*$ is not a level-0 individual in h', then (ii) must hold. But then the immediate supervisor of $\phi(i^*) = i^*$ in h' is $p_{h'}^2(i^*)$, which links (via ϕ^{-1}) to $p_h^2(i^*)$ in h, a supervisor of i^* in h, implying that (*) holds.

Since $h \succeq_H h'$, it remains to show that $h' \succeq_H h$. Proceeding by contradiction, $h' \succeq_H h$ implies that $h' \sim_H h$. Consequently, h' is a relabeling of h (Lemma 4), contradicting that h' can be obtained from h by removing a subordination relation.

Lemma 6. Suppose that $h, h' \in \mathcal{H}_n$, and let I_0 (resp., I'_0) be the set of level-0 individuals in h (resp., h'). The following two statements are equivalent:

- (i) $h \succeq_H h'$.
- (ii) There exists a finite partition of I'_0 consisting of $\#I_0$ elements,

$$\{I'_1,\ldots,I'_{\#I_0}\},\$$

where $\#I_0$ denotes the cardinality of I_0 , such that for each $i \in I_0$, $h(i) \geq_H (h'(\iota))_{\iota \in I'_i}$.

Proof. Suppose that (ii) holds. Then there exists a finite partition of I'_0 consisting of $#I_0$ elements,

$$\{I'_1, \ldots, I'_{\#I_0}\},$$
 (10)

such that for each $i \in I_0$, there exists a bijection ϕ_i from the set of individuals in h(i) to the set of individuals in $(h'(\iota))_{\iota \in I'_i}$ satisfying the following: for each individual j in h(i) such that $\phi_i(j)$ is not a level-0 individual,

$$\phi_i^{-1}(p_{(h'(l))_{i \in l'_i}}(\phi_i(j))) = p_{h(i)}^l(j), \quad \text{for some } l.$$
(11)

Define a function ϕ from the set of individuals in *h* to the set of individuals in *h'* as follows:

$$\phi(j) = \phi_i(j) \quad \text{if } j \in h(i), i \in I_0.$$

We claim that ϕ is a bijection. To see this, note first that if j' is an individual in h' then j' is an individual in the sub-hierarchy h'(i') for some $i' \in I'_0$. Since $i' \in I'_0$, there exists $i \in I_0$ such that $i' \in I'_i$. Thus, j' is an individual in the sub-hierarchy $(h'(\iota))_{\iota \in I'_i}$, and so there exists an individual j in the sub-hierarchy h(i) such that $j = \phi_i^{-1}(j')$. Because $i \in I_0$ and $j \in h(i)$, we see that

$$\phi(j) = \phi_i(j) = \phi_i(\phi_i^{-1}(j')) = j'.$$

Hence, ϕ is an onto map.

To see that ϕ is one-to-one, note that for every j in h, there is a unique $i \in I_0$ such that j is an individual in the sub-hierarchy h(i), and so there is a unique element I'_i of the partition in (10) and a unique individual i' in the sub-hierarchy $(h'(\iota))_{\iota \in I'_i}$ such that $\phi(j) = i'$.

Thus, ϕ is a bijection.

Next, fix an arbitrary individual j in h such that $\phi(j)$ is not a level-0 individual. Then $j \in h(i)$ for some $i \in I_0$ and $\phi(j) = \phi_i(j)$, where ϕ_i is a bijection from the set of individuals in h(i) to the set of individuals in $(h'(\iota))_{\iota \in I'_i}$ satisfying (11). Therefore, since

$$\phi^{-1}(p_{h'}(\phi(j))) = \phi^{-1}(p_{h'}(\phi_i(j))) = \phi^{-1}(p_{(h'(\iota))_{\iota \in I'_i}}(\phi_i(j))) = \phi_i^{-1}(p_{(h'(\iota))_{\iota \in I'_i}}(\phi_i(j)))$$

and

$$p_h^l(j) = p_{h(i)}^l(j),$$

we have

$$\phi^{-1}(p_{h'}(\phi_i(j))) = p_h^l(j), \text{ for some } l,$$

and so (i) holds.

Now suppose that (i) holds. Then, there exists a bijection ϕ from the set of individuals in *h* to the set of individuals in *h*' satisfying the following:

(*) For each individual *i* in *h* such that $\phi(i)$ is not a level-0 individual,

$$\phi^{-1}(p_{h'}(\phi(i))) = p_h^l(i), \text{ for some } l.$$

For each $i \in I_0$ (resp., $i \in I'_0$), let $I_{h(i)}$ (resp., $I_{h'(i)}$) be the set of all individuals in h(i) (resp., h'(i)).

First, we show that

$$\forall i \in I'_0, \exists j \in I_0 : \phi^{-1}(I_{h'(i)}) = \{\phi^{-1}(\iota) : \iota \in I_{h'(i)}\} \subseteq I_{h(j)}.$$
(12)

Fix $i \in I'_0$. Then $\phi^{-1}(i)$ is an individual in h. Let j be the level-0 supervisor of $\phi^{-1}(i)$ in h. It suffices to show that $\phi^{-1}(I_{h'(i)}) \subseteq I_{h(j)}$.

Proceeding by contradiction, suppose that there exists $\iota \in \phi^{-1}(I_{h'(i)}) \setminus I_{h(j)}$. Then $\iota \in I_{h(\iota^*)}$ for some $\iota^* \in I_0 \setminus \{j\}$. Note that $\phi(\iota) \neq i$, since $\phi^{-1}(i) \neq \iota$. Since *i* is the only level-0 individual in h'(i), and since $I_{h'(i)} \ni \phi(\iota) \neq i$, $\phi(\iota)$, an individual in the sub-hierarchy h'(i), is not a level-0 individual. Therefore, by (*),

$$\phi^{-1}(p_{h'}(\phi(\iota))) = p_h^l(\iota), \text{ for some } l,$$

implying that

$$\phi^{-1}(p_{h'}(\phi(\iota))) \in I_{h(\iota^*)}.$$
(13)

If $p_{h'}(\phi(\iota))$ is a level-0 individual, since $p(\phi(\iota)) \in I_{h'(i)}$, then $p(\phi(\iota)) = i$ (since *i* is the only level-0 individual in h'(i)); in this case, since $\phi^{-1}(i) \in I_{h(j)}$ and $j \neq \iota^*, \phi^{-1}(p_{h'}(\phi(\iota))) = \phi^{-1}(i)$ cannot be a member of $I_{h(\iota^*)}$, contradicting (13).

If $p(\phi(\iota))$ is not a level-0 individual, then, again applying (*), we see that

$$\phi^{-1}(p_{h'}^2(\phi(\iota))) = p_h^l(\iota), \text{ for some } l,$$

implying that

$$\phi^{-1}(p_{h'}^2(\phi(\iota))) \in I_{h(\iota^*)}.$$
(14)

If $p_{h'}^2(\phi(\iota))$ is a level-0 individual, since $p_{h'}^2(\phi(\iota)) \in I_{h'(i)}$, then $p_{h'}^2(\phi(\iota)) = i$; in this case, since $\phi^{-1}(i) \in I_{h(j)}$ and $j \neq \iota^*$, $\phi^{-1}(p_{h'}^2(\phi(\iota))) = \phi^{-1}(i)$ cannot be a member of $I_{h(\iota^*)}$, which contradicts (14).

If $p^2(\phi(\iota))$ is not a level-0 individual, again applying (*), we see that

$$\phi^{-1}(p_{h'}^3(\phi(\iota))) = p^l(\iota), \text{ for some } l,$$

implying that $\phi^{-1}(p_{h'}^3(\phi(\iota))) \in I_{h(\iota^*)}$. This argument can be reiterated until a contradiction is reached in finitely many steps.

This proves (12).

Next, we show that there exists a finite partition of I'_0 consisting of $\#I_0$ elements,

$$\{I'_1,\ldots,I'_{\#I_0}\},$$

such that for each $i \in I_0$, we have

$$\phi(I_{h(i)}) = \{\phi(\iota) : \iota \in I_{h(i)}\} = \bigcup_{\iota \in I'_i} I_{h'(\iota)}.$$
(15)

To see this, note that by (12), for each $\iota \in I'_0$, there exists $j_\iota \in I_0$ such that

$$\phi^{-1}(I_{h'(\iota)}) = \{\phi^{-1}(\iota') : \iota' \in I_{h'(\iota)}\} \subseteq I_{h(j_{\iota})}.$$

In addition, because ϕ is a bijection, each j_t must be unique.

For $i \in I_0$, define

$$I'_{i} = \{\iota \in I'_{0} : j_{\iota} = i\}.$$
(16)

First, we show that

$$\{I'_1,\ldots,I'_{\#I_0}\},\$$

where each I'_i is defined by (16), is a partition of I'_0 . First, note that for $i, j \in I_0$ with $i \neq j$,

$$I'_{i} \cap I'_{j} = \{\iota \in I'_{0} : j_{\iota} = i\} \cap \{\iota \in I'_{0} : j_{\iota} = j\} = \emptyset,$$

since j_{ι} is uniquely defined for each $\iota \in I'_0$. Next, note that

$$\bigcup_{i\in I_0} I'_i = I'_0$$

To see this, note that the containment $\bigcup_{i \in I_0} I'_i \subseteq I'_0$ is obvious, so we only need to show that $\bigcup_{i \in I_0} I'_i \supseteq I'_0$.

Suppose that $\iota \in I'_0$. Then $\iota \in I'_{j_i}$, and so $\iota \in \bigcup_{i \in I_0} I'_i$. Thus, $\bigcup_{i \in I_0} I'_i \supseteq I'_0$.

Next, we prove (15) for each $i \in I_0$.

Fix $i \in I_0$. Suppose that $j' \in \phi(I_{h(i)})$. Then there exists $j \in I_{h(i)}$ such that $j' = \phi(j)$, implying that $j' \in I_{h'(\iota)}$ for some $\iota \in I'_0$. If $\iota \notin I'_i$, then there exists $j_\iota \in I_0 \setminus \{i\}$ such that $\phi^{-1}(I_{h'(\iota)}) \subseteq I_{h(j_\iota)}$. Since $j' \in I_{h'(\iota)}$, this implies that

$$j = \phi^{-1}(j') \in I_{h(j_l)}.$$

But this contradicts the fact that $j \in I_{h(i)}$. Indeed, since $j_i \neq i$, we have

$$I_{h(j_i)} \cap I_{h(i)} = \emptyset.$$

Therefore, we must have $j' \in I_{h'(\iota)}$ for some $\iota \in I'_i$.

Hence, $\phi(I_{h(i)}) \subseteq \bigcup_{\iota \in I'_i} I_{h'(\iota)}$.

Conversely, if $j' \in I_{h'(\iota)}$ for some $\iota \in I'_i$, then the definition of I'_i in (16) entails that

$$\phi^{-1}(I_{h'(\iota)}) \subseteq I_{h(i)},$$

implying that $\phi^{-1}(j') \in I_{h(i)}$, and so $j' \in \phi(I_{h(i)})$.

Consequently, $\phi(I_{h(i)}) \supseteq \bigcup_{\iota \in I'_i} I_{h'(\iota)}$.

We conclude that (15) holds. Now let $\phi|_{I_{h(i)}}$ be the restriction of ϕ to $I_{h(i)}$.

By (15), $\phi|_{I_{h(i)}}$ is a bijection from $I_{h(i)}$ to $\bigcup_{\iota \in I'_i} I_{h'(\iota)}$.

By (*), $\phi|_{I_{h(i)}}$ satisfies the following: for each individual j in h(i) such that $\phi|_{I_{h(i)}}(j)$ is not a level-0 individual in $(h'(\iota))_{\iota \in I'_{i}}$,

$$\phi|_{I_{h(i)}}^{-1}(p_{(h'(l))_{l\in I'_{i}}}(\phi|_{I_{h(i)}}(j))) = p^{l}_{h(i)}(j), \text{ for some } l.$$

Lemma 7. Suppose that $h, h' \in \mathcal{H}_n$, and let I_0 (resp., I'_0) be the set of level-0 individuals in h (resp., h'). Then (i) implies (ii):

- (i) $h \succ_H h'$.
- (ii) For each $i \in I_0$, there exists $I'_i \subseteq I'_0$ such that $h(i) \succeq_H (h'(j))_{j \in I'_i}$; and there exists $i^* \in I_0$ such that $h(i^*) \succ_H (h'(j))_{j \in I'_*}$.

Proof. Suppose that $h >_H h'$. Then $h \succcurlyeq_H h'$, and so Lemma 6 implies that there exists a finite partition of I'_0 consisting of $\#I_0$ elements,

$$\{I'_1,\ldots,I'_{\#I_0}\},\$$

where $\#I_0$ denotes the cardinality of I_0 , such that for each $i \in I_0$, $h(i) \succeq_H (h'(\iota))_{\iota \in I'_i}$.

If $(h'(\iota))_{\iota \in I'_i} \succeq_H h(i)$ for each $i \in I_0$, then $h(i) \sim_H (h'(\iota))_{\iota \in I'_i}$ for each $i \in I_0$, and so by Lemma 4, h(i) is a relabeling of $(h'(\iota))_{\iota \in I'_i}$ for each $i \in I_0$. But then

$$h = (h(i))_{i \in I_0}$$
 and $h' = ((h'(\iota))_{\iota \in I'_i})_{i \in I_0}$

are relabelings of each other, and so A implies that

$$h = (h(i))_{i \in I_0} \sim_H h' = ((h'(\iota))_{\iota \in I'_i})_{i \in I_0},$$

a contradiction (recall that \succeq_H satisfies A by Proposition 1).

Therefore, there exists $i^* \in I_0$ such that $(h'(\iota))_{\iota \in I'_{i^*}} \not\geq_H h(i^*)$, implying that $h(i^*) \succ_H (h'(\iota))_{\iota \in I'_{i^*}}$, and so (ii) holds.

A.2. Proof of Lemma 1

Lemma 1 is restated here for the reader's convenience.

Lemma 1. The hierarchical pre-order \succeq_H defined on \mathcal{H}_n is reflexive and transitive and satisfies A and SR.

Proof. Reflexivity follows immediately from the definition of \succeq_H .

Let $I_{\tilde{h}}$ represent the set of individuals in $h \in \mathcal{H}_n$.

To see that \geq_H is transitive, suppose that

$$h \succeq_H h' \succeq_H h'', \text{ for } h, h', h'' \in \mathscr{H}_n$$

Then, there exist bijections $\phi : I_h \to I_{h'}$ and $\phi' : I_{h'} \to I_{h''}$ satisfying the following:

{1} For each individual *i* in *h* such that $\phi(i)$ is not a level-0 individual, the immediate supervisor of $\phi(i)$ in *h'*, $p_{h'}(\phi(i))$, links (via ϕ^{-1}) to a supervisor *j* of *i* in *h*, i.e.,

$$\phi^{-1}(p_{h'}(\phi(i))) = j = p_h^l(i)$$
, for some *l*.

{2} For each individual *i* in *h*' such that $\phi'(i)$ is not a level-0 individual, the immediate supervisor of $\phi'(i)$ in *h*", $p_{h''}(\phi'(i))$, links (via ϕ'^{-1}) to a supervisor *j* of *i* in *h*', i.e.,

$$\phi'^{-1}(p_{h''}(\phi'(i))) = j = p_{h'}^{l}(i), \text{ for some } l.$$

Since ϕ and ϕ' are bijections, the composition $\phi^* := \phi' \circ \phi$ is also a bijection (see, e.g., Blyth, 1975, Theorem 5.10, p. 37). Thus, it suffices to show the following:

(•) For each individual *i* in *h* such that $\phi^*(i)$ is not a level-0 individual, the immediate supervisor of $\phi^*(i)$ in *h*", $p_{h''}(\phi^*(i))$, links (via ϕ^{*-1}) to a supervisor *j* of *i* in *h*, i.e.,

$$\phi^{*-1}(p_{h''}(\phi^{*}(i))) = j = p_{h}^{l}(i), \text{ for some } l.$$

Fix an individual *i* in *h* such that $\phi^*(i)$ is not a level-0 individual. Proceeding by contradiction, suppose that

$$\phi^{*-1}(p_{h''}(\phi^*(i))) \neq p_h^l(i), \text{ for any } l.$$
 (17)

Consider the sequence

$$i, \phi(i), p_{h'}(\phi(i)), \phi^{-1}(p_{h'}(\phi(i))), \phi^{-1}(p_{h'}(\phi(i))), p_{h'}(\phi(i)), p_{h'}^2(\phi(i)), \phi^{-1}(p_{h'}^2(\phi(i))), \phi^{-1}(p_{h'}^2(\phi(i))), p_{h'}^2(\phi(i)), p_{h'}^3(\phi(i)), \phi^{-1}(p_{h'}^3(\phi(i))), \dots$$

This sequence can be subdivided into four-element cycles as follows:

Cycle 1:
$$i, \phi(i), p_{h'}(\phi(i)), \phi^{-1}(p_{h'}(\phi(i))).$$

Cycle 2: $\phi^{-1}(p_{h'}(\phi(i))), p_{h'}(\phi(i)), p_{h'}^2(\phi(i)), \phi^{-1}(p_{h'}^2(\phi(i)))$
Cycle 3: $\phi^{-1}(p_{h'}^2(\phi(i))), p_{h'}^2(\phi(i)), p_{h'}^3(\phi(i)), \phi^{-1}(p_{h'}^3(\phi(i)))$
Cycle 4: $\phi^{-1}(p_{h'}^3(\phi(i))), p_{h'}^3(\phi(i)), p_{h'}^4(\phi(i)), \phi^{-1}(p_{h'}^4(\phi(i)))$
 \vdots \vdots

The first and last elements of each cycle are individuals in *h*, while the second and third elements of each cycle are individuals in *h*'. Moreover, by {1}, the first and last elements of each cycle belong to the path connecting *i* and *i*'s level-0 supervisor in *h*, i.e., if *j* is the first or the fourth element of a cycle, we have $j = p^{l}(i)$ for some *l*. In addition, by construction, the second and third elements of every cycle belong to the path connecting $\phi(i)$ and $\phi(i)$'s level-0 supervisor in *h*', i.e., if *j* is the second or the third element of a cycle, we have $j = p_{h'}^{l}(\phi(i))$ for some *l*.

Note that each individual in the path connecting $\phi(i)$ and $\phi(i)$'s level-0 supervisor in h' must eventually become the third element of a cycle. Hence, because

$$\phi'^{-1}(p_{h''}(\phi^*(i))) = p_{h'}^l(\phi(i)), \text{ for some } l \text{ (by {2})},$$

 $\phi'^{-1}(p_{h''}(\phi^*(i)))$ is equal to the third element of some cycle ℓ . But then the fourth element of cycle ℓ , which can be expressed as

$$\phi^{-1}(\phi'^{-1}(p_{h''}(\phi^*(i)))))$$

belongs to the path connecting i and i's level-0 supervisor in h (as noted in the previous paragraph). Noting that

$$\phi^{-1}(\phi'^{-1}(p_{h''}(\phi^*(i)))) = \phi^{*-1}(p_{h''}(\phi^*(i)))),$$

this contradicts our initial assumption in (17).

We conclude that (\circ) holds, implying that \succeq_H is transitive.

By Lemma 5, \succeq_H satisfies SR.

To see that \succeq_H satisfies **A**, let h' be a relabeling of h. Then there exists a bijection $\phi : I_h \to I_{h'}$ with the following property: if each individual i in h' is assigned the label " $\phi^{-1}(i)$," then the resulting hierarchy is identical to h.

It is easy to see that, for the bijection ϕ , the following condition is satisfied: for each individual *i* in *h* such that $\phi(i)$ is not a level-0 individual, the immediate supervisor of $\phi(i)$ in *h'*, $p_{h'}(\phi(i))$, links (via ϕ^{-1}) to a supervisor *j* of *i* in *h*, i.e.,

$$\phi^{-1}(p_{h'}(\phi(i))) = j = p_h^l(i), \text{ for some } l.$$

Hence, $h \succeq_H h'$.

A similar condition can be verified for the bijection $\phi^{-1} : I_{h'} \to I_h$: for each individual *i* in *h*' such that $\phi^{-1}(i)$ is not a level-0 individual, the immediate supervisor of $\phi^{-1}(i)$ in *h*, $p_h(\phi^{-1}(i))$, links (via ϕ) to a supervisor *j* of *i* in *h*', i.e.,

$$\phi(p_h(\phi^{-1}(i))) = j = p_{h'}^l(i)$$
, for some *l*.

Consequently, $h' \succeq_H h$ and $h \succeq_H h'$, implying that $h \sim_H h'$.

A.3. Proof of Theorem 1

Theorem 1. For $h, h' \in \mathcal{H}_n$, $h \succ_H h'$ if and only if h' can be obtained from some relabeling of h by successive removals of subordination relations.

Proof. [*Necessity.*] First, we prove the "only if" part of the statement under the assumption that *h* has only one level-0 individual.

We proceed by induction on *n*. The statement is clearly true if n = 1. We now prove the statement for any n > 1 under the assumption that it is true for *m*-person hierarchies, where $m \in \{1, ..., n - 1\}$.

Suppose that $h >_H h'$. We must show that h' can be obtained from some relabeling of h by successive removals of subordination relations.

Since $h \succeq_H h'$, there exists a bijection ϕ from the set of individuals in h to the set of individuals in h' satisfying the following:

(•) For each individual *i* in *h* such that $\phi(i)$ is not a level-0 individual, the immediate supervisor of $\phi(i)$ in *h'*, $p_{h'}(\phi(i))$, links (via ϕ^{-1}) to a supervisor *j* of *i* in *h*, i.e.,

$$\phi^{-1}(p_{h'}(\phi(i))) = j = p_h^l(i), \text{ for some } l.$$

Let the (unique) level-0 individual *h* be denoted by ι . Then $\phi(\iota)$ is a level-0 individual in *h*' (otherwise $p_{h'}(\phi(\iota))$) would not link (via ϕ^{-1}) to a supervisor of ι , contradicting (•)).

If $\phi(\iota) \neq \iota$, the individuals in *h* can be relabeled so that $\phi(\iota) = \iota$. The resulting relabeling will be denoted again by *h*.

Let $h \setminus \iota$ be the hierarchy resulting from removing individual ι from h: in $h \setminus \iota$, every j in the set S_{ι} of all level-1 subordinates of ι becomes a level-0 individual, and the sub-hierarchy that begins at j is h(j).

Let $h' \setminus \iota$ be the hierarchy resulting from removing individual ι from h':

- In h' \ ι, every j in the set of all level-1 subordinates of ι becomes a level-0 individual, and the sub-hierarchy that begins at j is h'(j).
- The structure of *h*′ remains otherwise intact, i.e., the sub-hierarchy that begins at any level-0 *i* other than *ι* is *h*′(*i*).

Let ϕ^* be the restriction of ϕ to the individuals in $h \setminus \iota$. Note that ϕ^* is a bijection between the individuals in $h \setminus \iota$ and those in $h' \setminus \iota$. Moreover, because ϕ satisfies (•) and $\phi(\iota) = \iota, \phi^*$ has the following property: for each individual i in $h \setminus \iota$ such that $\phi^*(i)$ is not a level-0 individual,

$$\phi^{*-1}(p_{h'}(\phi^{*}(i))) = p_{h}^{l}(i), \text{ for some } l.$$

Thus, we have $h \setminus \iota, h' \setminus \iota \in \mathscr{H}_{n-1}$ and $h \setminus \iota \succeq_H h' \setminus \iota$.

Suppose first that $h' \setminus \iota \succeq_H h \setminus \iota$. Then, $h \setminus \iota \sim_H h' \setminus \iota$, and Lemma 4 implies that $h \setminus \iota$ is a relabeling of $h' \setminus \iota$.

Since ι is the only level-0 individual in h, we can write

$$h = h(\iota)$$
 and $h' = (h'(\iota), (h'(j))_{j \in I'_0 \setminus \{\iota\}}),$

where I'_0 denotes the set of all level-0 individuals in h'. Now, letting S_i (resp., S'_i) be the set of level-1 subordinates of i in h (resp., h'), we can write

$$h \setminus \iota = (h(j))_{j \in S_{\iota}}$$
 and $h' \setminus \iota = ((h'(j))_{j \in S'_{\iota}}, (h'(j))_{j \in I'_{0} \setminus \{\iota\}})$

Since $h \setminus \iota$ is a relabeling of $h' \setminus \iota$, there is no loss of generality in assuming that $h \setminus \iota$ and $h' \setminus \iota$ are identical (since the individuals in $h \setminus \iota$ can always be relabeled in such a way that $h \setminus \iota$ and $h' \setminus \iota$ are identical). Hence,

$$S_{\iota} = S'_{\iota} \cup (I'_0 \setminus \{\iota\}) \quad \text{and} \quad h(j) = h'(j) \text{ for all } j \in S_{\iota}.$$

$$(18)$$

Now let $\{i_1, \ldots, i_m\}$ be an enumeration of $I'_0 \setminus \{\iota\}$ and define the sequence of hierarchies h_0, \ldots, h_m as follows:

- $h_0 = h$.
- *h*₁ is obtained from *h* by the removal of a subordination relation as follows: *i*₁ is no longer a level-1 subordinate in *h* under the direct direct supervision of *ι*, but rather a level-0 individual, and the sub-hierarchy that begins at *i*₁ is *h*(*i*₁); *h*₁ is otherwise equal to *h*.
- *h*₂ is obtained from *h*₁ by removing a subordination relation as follows: *i*₂ is no longer a level-1 subordinate in *h*₁ under the direct supervision of *ι*, but rather a level-0 individual, and the sub-hierarchy that begins at *i*₂ is *h*(*i*₂); *h*₂ is otherwise equal to *h*₁.

:

h_m is obtained from *h_{m-1}* by removing a subordination relation as follows: *i_m* is no longer a level-1 subordinate in *h_{m-1}* under the direct supervision of *ι*, but rather a level-0 individual, and the sub-hierarchy that begins at *i_m* is *h*(*i_m*); *h_m* is otherwise equal to *h_{m-1}*.

Next, suppose that $h' \setminus \iota \not\geq_H h \setminus \iota$. Since $h \setminus \iota \succcurlyeq_H h' \setminus \iota$, we see that $h \setminus \iota \succ_H h' \setminus \iota$. Since $h \setminus \iota, h' \setminus \iota \in \mathscr{H}_{n-1}$ and $h \setminus \iota \succ_H h' \setminus \iota$, the induction hypothesis gives some relabeling of $h \setminus \iota$, denoted again by $h \setminus \iota$, such that

$$h' \setminus \iota \Leftarrow_{RS} h \setminus \iota;$$

here (and in the remainder of the proof), for any two hierarchies \hat{h} and \overline{h} , " $\hat{h} \leftarrow_{RS} \overline{h}$ " means that " \hat{h} can be obtained from \overline{h} by successive removals of subordination relations."

Recall that h and h' can be expressed as

$$h = h(\iota)$$
 and $h' = (h'(\iota), (h'(j))_{j \in I'_0 \setminus \{\iota\}}),$

and that $h \setminus \iota$ and $h' \setminus \iota$ are expressible as

$$h \setminus \iota = (h(j))_{j \in S_{\iota}}$$
 and $h' \setminus \iota = ((h'(j))_{j \in S'_{\iota}}, (h'(j))_{j \in I'_{0} \setminus \{\iota\}}),$

where S_{ι} (resp., S'_{ι}) represents the set of level-1 subordinates of ι in h (resp., h').

Because every removal of a subordination relation in the transition

$$h' \setminus \iota \Leftarrow_{RS} h \setminus \iota$$

affects only the players of one and only one of the sub-hierarchies in $(h(j))_{j \in S_i}$, there exists a partition

$$(I_j)_{j \in S}$$

of the set $S'_{\iota} \cup (I'_0 \setminus \{\iota\})$ such that each I_j is a subset of the set of individuals in h(j) and

$$(h'(j'))_{j' \in I_j} \Leftarrow_{RS} h(j), \text{ for all } j \in S_\iota.$$
 (19)

Each partition member I_j can be further partitioned into two sets: the members of I_j that are immediate subordinates of ι in h', I_j^s , and the members of I_j that are not immediate subordinates of ι in h', I_j^{ns} :

$$I_j^s = I_j \cap S'_\iota$$
 and $I_j^{ns} = I_j \cap (I'_0 \setminus \{\iota\})$

Using this notation, (19) can be rewritten as

$$((h'(j'))_{j' \in I_i^s}, (h'(j'))_{j' \in I_i^{ns}}) \Leftarrow_{RS} h(j), \quad \text{for all } j \in S_\iota.$$

$$(20)$$

Now let h'' be the hierarchy obtained from the hierarchy

$$((h'(j'))_{j'\in I_j^s}, (h'(j'))_{j'\in I_j^{ns}})_{j\in S_d}$$

by adding individual ι at the top, so that ι is the only level-0 individual in h'' and the level-1 subordinates of ι are the members of

$$S'_{\iota} \cup (I'_{0} \setminus \{\iota\}) = \bigcup_{j \in S_{\iota}} I_{j} = \left(\bigcup_{j \in S_{\iota}} I^{s}_{j}\right) \cup \left(\bigcup_{j \in S_{\iota}} I^{ns}_{j}\right).$$

Similarly, let h^* be the hierarchy obtained from

$$(h(j))_{j\in S_{\iota}}$$

by adding individual ι at the top, so that ι is the only level-0 individual in h^* and the level-1 subordinates of ι are the members of S_{ι} .

Note that (20) implies that

$$h'' \Leftarrow_{RS} h^*$$
.

Consequently, since $h^* = h = h(\iota)$, we have

$$h'' \Leftarrow_{RS} h(\iota) = h. \tag{21}$$

Note that, by successive removals of subordination relations in h'', we can, for any level-1 subordinate j' in $\bigcup_{j \in S_i} I_j^{ns}$, move the sub-hierarchy h'(j') up to level 0, thus obtaining the hierarchy

$$(h''', (h'(j'))_{j' \in \bigcup_{j \in S_l} I_j^{ns}}),$$

where h''' is a hierarchy defined as follows:

- *h*^{*m*} has only one level-0 individual, *ι*.
- The level-1 subordinates of *ι* are the members of ∪_{*j*∈*S_ι} <i>I^s_j*, and the sub-hierarchy that begins at any such level-1 subordinate *j'* is given by *h'*(*j'*).
 </sub>

We therefore have

$$(h''', (h'(j'))_{j' \in \bigcup_{i \in S_l} I_i^{ns}}) \Leftarrow_{RS} h''.$$

$$(22)$$

Moreover, since all the immediate subordinates of ι in h' are the same as all the immediate subordinates of ι in h''' and all the non-subordinates of ι in h' are the same as all the non-subordinates of ι in h' are the same as all the non-subordinates of ι in

$$(h''', (h'(j'))_{j' \in \bigcup_{j \in S_l} I_j^{ns}}),$$

we see that

$$h' = (h''', (h'(j'))_{j' \in \bigcup_{j \in S_i} I_i^{ns}}).$$

This, together with (21)-(22), gives $h' \leftarrow_{RS} h$, as desired.

It remains to prove the "only if" part of the statement when *h* has more than one level-0 individual.

Suppose that $h \succ_H h'$. We must show that h' can be obtained from some relabeling of h by successive removals of subordination relations.

Let I_0 (resp., I'_0) be the set of level-0 individuals in h (resp., h'). By Lemma 7, for each $i \in I_0$, there exists $I'_i \subseteq I'_0$ such that $h(i) \succeq_H (h'(j))_{j \in I'_i}$; and there exists $i^* \in I_0$ such that $h(i^*) \succ_H (h'(j))_{j \in I'_i}$.

From the first part of this proof, we obtain the following:

$$(h'(j))_{j \in I'_i} \Leftarrow_{RS} h(i)$$
, for all $i \in I^*$.

Therefore, since $h(i) \sim_H (h'(j))_{j \in I'_i}$ for each $i \in I_0 \setminus I^*$, and since the relation $h(i) \sim_H (h'(j))_{j \in I'_i}$ implies that $(h'(j))_{j \in I'_i}$ is a relabeling of h(i) (Lemma 4), it follows that h' can be obtained from some relabeling of h by successive removals of subordination relations.

[*Sufficiency*.] Suppose that h' can be obtained from some relabeling of h, denoted by \overline{h} , by successive removals of subordination relations, i.e.,

$$h' \leftarrow_{RS} h_1 \leftarrow_{RS} \cdots \leftarrow_{RS} h_L \leftarrow_{RS} \overline{h}$$

for finitely many hierarchies h_1, \ldots, h_L ; here (and in the remainder of the proof), for any two hierarchies \hat{h} and \underline{h} , " $\hat{h} \leftarrow_{RS} \underline{h}$ " means that " \hat{h} can be obtained from \underline{h} by removing a subordination relation." We must show that $h >_H h'$.

By Lemma 5,

 $\overline{h} \succ_H h_L \succ_H \cdots \succ_H h_1 \succ_H h'.$

By reflexivity and transitivity of \succeq_H (Lemma 1), it follows that $\overline{h} \succeq_H h'$ (Sen, 2017, Lemma 1*a, p. 56). Moreover, since \overline{h} is a relabeling of h, Lemma 1 gives $h \sim_H \overline{h}$. Consequently,

$$h \sim_H \overline{h} \succ_H h'$$
,

implying that $h >_H h'$ (Sen, 2017, Lemma 1*a, p. 56).

A.4. Proof of Proposition 1

Proposition 1. *The hierarchical pre-order* \geq_s *defined on* \mathcal{H}_n *is reflexive and transitive and satisfies* **A** *and* **SR***.*

Proof. Reflexivity follows immediately from the definition of \succeq_s .

To see that \succeq_s is transitive, suppose that

$$h \geq_s h' \geq_s h''$$
, for $h, h', h'' \in \mathcal{H}_n$.

Then, letting $I_{\hat{h}}$ represent the set of individuals in hierarchy \hat{h} , there exist bijections $\phi : I_h \to I_{h'}$ and $\phi' : I_{h'} \to I_{h''}$ satisfying the following:

- For each individual *i* in *h*, the number of supervisors of *i* in *h*, #_h*i*, is greater than or equal to the number of supervisors of φ(*i*) in h', #_{h'}φ(*i*).
- For each individual *i* in *h'*, the number of supervisors of *i* in *h'*, #_{h'}*i*, is greater than or equal to the number of supervisors of φ'(*i*) in *h*", #_{h"}φ'(*i*).

Since ϕ and ϕ' are bijections, the composition $\phi^* := \phi' \circ \phi$ is also a bijection (see, e.g., Blyth, 1975, Theorem 5.10, p. 37). Moreover, for each individual *i* in *h*, we have

$$\#_h i \ge \#_{h'} \phi(i) \ge \#_{h''} \phi'(\phi(i)).$$

Consequently, for each individual i in h,

$$\#_{h}i \geq \#_{h''}[\phi' \circ \phi](i) = \#_{h''}\phi^{*}(i),$$

implying that $h \succeq_s h''$.

To see that \succeq_s satisfies **A**, suppose that h' is a relabeling of h. Then there exists a bijection $\phi : I_h \to I_{h'}$ with the following property: if each individual i in h' is assigned the label " $\phi^{-1}(i)$," then the resulting hierarchy is identical to h.

For the bijection ϕ , the following condition is satisfied: for each *i* in *h*, the number of supervisors of *i* in *h* is equal to the number of supervisors of $\phi(i)$ in *h*'.

Hence, $h \succeq_s h'$.

A similar condition can be verified for the bijection $\phi^{-1} : I_{h'} \to I_h$: for each individual *i* in *h*', the number of supervisors of *i* in *h*' is equal to the number of supervisors of $\phi^{-1}(i)$ in *h*.

Consequently, $h' \geq_s h$ and $h \geq_s h'$, implying that $h \sim_s h'$.

It remains to show that \succeq_s satisfies SR. Suppose that h' can be obtained from h by removing a subordination relation. Then there exists a level-k subordinate i^* in h, where k > 0, satisfying the following:

- (i) If i^* 's immediate supervisor in h, $p_h(i^*)$, is a level-0 individual, then h' is the hierarchy in which the sub-hierarchy $h(i^*)$ is no longer under $p_h(i^*)$'s supervision, i^* becomes a level-0 individual, and the sub-hierarchy that begins at i^* is $h(i^*)$; h' is otherwise equal to h.
- (ii) If *i**'s immediate supervisor in *h*, $p_h(i^*)$, is a not level-0 individual, then $p_h(i^*)$ is an immediate subordinate of $p_h^2(i^*)$, i.e., $p_h(i^*) \in S_{p_h^2(i^*)}$. In this case, *h*' is the hierarchy in which the sub-hierarchy $h(i^*)$ is no longer under $p_h(i^*)$'s supervision, but rather under the direct supervision of individual $p_h^2(i^*)$, so that i^* is no longer a level-*k* subordinate, but rather a level-(k 1) subordinate in $S_{p_h^2(i^*)}$, and the sub-hierarchy that begins at i^* is $h(i^*)$; *h*' is otherwise equal to *h*.

We must show that $h \succ_s h'$.

Note that the only individuals whose set of supervisors is altered as a result of the subordination removals specified in items (i) and (ii) are those in the sub-hierarchy $h(i^*)$ of h containing i^* and all of i^* 's subordinates. Moreover, after the removal of a subordination relation, the individuals in $h(i^*)$ are left with less supervisors. Consequently, if i is an individual in h not in the sub-hierarchy $h(i^*)$, the number of supervisors of i in h is equal to the number of supervisors of i in h', while if i is in $h(i^*)$, the number of supervisors of i in h is greater than the number of supervisors of i in h', implying that $h \succeq_s h'$.

It remains to show that $h' \not\geq_s h$. Proceeding by contradiction, suppose that $h' \not\geq_s h$. Then there exists a bijection $\varphi : I_{h'} \to I_h$ such that for each i in h', the number of supervisors of i in h' is greater than or equal to the number of supervisors of $\varphi(i)$ in h.

Let I^* be the (nonempty) set of all individuals in $h(i^*)$ who have the most supervisors in h among all the individuals in $h(i^*)$. Let s^* be the number of supervisors in h for the individuals in I^* . Then $s^* - 1$ is the number of supervisors in h' for the individuals in I^* (since h' can be obtained from h by removing a subordination relation and (i) and (ii) hold).

Let \overline{s} be the maximum number of supervisors that an individual in h can have. Note that $\overline{s} \ge s^*$.

We claim that if $\overline{s} > s^*$, then $\varphi(I'_{\overline{s}}) = I_{\overline{s}}$.

To see this, note that if $\overline{s} > s^*$, then all the individuals in *h* with \overline{s} supervisors are not in the sub-hierarchy $h(i^*)$, and so the number of individuals in *h* with \overline{s} supervisors, denoted by $I_{\overline{s}}$, is equal to the number of individuals in *h*' with \overline{s} supervisors, denoted by $I'_{\overline{s}}$.

This implies that $\varphi(I'_{\overline{s}}) \supseteq I_{\overline{s}}$. Indeed, if that were not the case, there would exist an individual $\iota \in I_{\overline{s}} \setminus \varphi(I'_{\overline{s}})$, i.e., ι would have \overline{s} supervisors in h and any individual in h' with \overline{s} supervisors would link, via φ , to an individual in h other than ι . But since $I_{\overline{s}} = I'_{\overline{s}}$, this would imply that for some individual in h' with less than \overline{s} supervisors, $\iota', \varphi(\iota') = \iota$, contradicting the fact that for each i in h', the number of supervisors of i in h' is greater than or equal to the number of supervisors of $\varphi(i)$ in h.

Since $I_{\overline{s}} = I'_{\overline{s}}$ and φ is a bijection, $\varphi(I'_{\overline{s}})$ and $I_{\overline{s}}$ have the same cardinality, and so the containment $\varphi(I'_{\overline{s}}) \supseteq I_{\overline{s}}$ implies that $\varphi(I'_{\overline{s}}) = I_{\overline{s}}$.

Similarly, for any $\ell \in \mathbb{N}$ for which $\overline{s} - \ell > s^*$, we have $\varphi(I'_{\overline{s}-\ell}) = I_{\overline{s}-\ell}$ (where $I_{\overline{s}-\ell}$ (resp., $I'_{\overline{s}-\ell}$) represents the set of all individuals in h (resp., h') with $\overline{s} - \ell$ supervisors).

Next, note that there exists $\ell^* \in \{0, 1, 2, ...\}$ such that $\overline{s} - \ell^* = s^*$, since $\overline{s} \ge s^*$. Therefore, $I_{\overline{s}-\ell^*} = I_{s^*}$. Moreover, since I_{s^*} is the set of all individuals in h with s^* supervisors, and since I^* is the set of all individuals in the sub-hierarchy $h(i^*)$ of h with s^* supervisors, it follows that I_{s^*} contains I^* .

Next, we show that $I'_{s^*} = I_{s^*} \setminus I^*$. To see this, suppose that $i \in I'_{s^*}$. Then *i* is not in $h(i^*)$. Indeed, if *i* were in $h(i^*)$, since *i* has s^* supervisors in h', then *i* would have $s^* + 1$ supervisors in *h*, contradicting the fact that those individuals in $h(i^*)$ who have the most supervisors in *h* have s^* supervisors.

Since *i* is not in $h(i^*)$, we have $i \notin I^*$ (since the members of I^* are also in $h(i^*)$). Now, because h' can be obtained from h by removing a subordination relation and (i) and (ii) hold, and since the removal of a subordination relation specified in items (i) and (ii) does not affect the number of supervisors for those individuals not in $h(i^*)$, $i \in I'_{s^*}$ implies $i \in I_{s^*}$. Thus, $i \in I_{s^*} \setminus I^*$, and so $I'_{s^*} \subseteq I_{s^*} \setminus I^*$.

Conversely, suppose that $i \in I_{s^*} \setminus I^*$. Then *i* is not in $h(i^*)$ (since I^* is the set of all individuals in $h(i^*)$ who have s^* supervisors in *h*). Therefore, because the removal of a subordination relation specified in items (i) and (ii) does not affect the number of supervisors for those individuals not in $h(i^*)$, we have $i \in I'_{s^*}$. Hence, $I'_{s^*} \supseteq I_{s^*} \setminus I^*$.

Now, since $I'_{s^*} = I_{s^*} \setminus I^*$ and I^* is nonempty, it follows that the number of individuals in *h* with *s*^{*} supervisors exceeds the number of individuals in *h'* with *s*^{*} supervisors. Consequently, using the fact (proven earlier) that

$$\varphi(I'_{\overline{s}-\ell}) = I_{\overline{s}-\ell}, \text{ for any } \ell \in \mathbb{N} \text{ for which } \overline{s}-\ell > s^*,$$

we see that there exists some individual ι in h' with less than s^* supervisors whose corresponding individual in h, $\varphi(\iota)$, has s^* supervisors. But this contradicts the fact that the number of supervisors of ι in h' is greater than or equal to the number of supervisors of $\varphi(\iota)$ in h.

We conclude that $h' \not\geq_s h$. Since $h \succ_s h'$ and $h' \not\geq_s h$, we see that $h \succ_p h'$.

References

- Bichler, Shimshon and Jonathan Nitzan (2020) "Growing through Sabotage: Energizing Hierarchical Power," *Review of Capital as Power*, Vol. 5, pp. 1–78.
- Blyth, T. S. (1975) Set Theory and Abstract Algebra, London, New York: Longman.
- Bossert, Walter, Burak Can, and Conchita D'Ambrosio (2016) "Measuring rank mobility with variable population size," *Social Choice and Welfare*, Vol. 46, pp. 917–931.
- Chakravarty, S. R. (2009) *Inequality, Polarization and Poverty*: Springer Netherlands, pp.1–178, DOI: http://dx.doi.org/10.1007/978-0-387-79253-8.
- (2015) Inequality, Polarization and Conflict: An Analytical Study, Economic Studies in Inequality, Social Exclusion and Well-Being: Springer, DOI: http://dx.doi.org/10. 1007/978-81-322-2166-1.
- Chakravarty, Satya R. and Conchita D'Ambrosio (2013) "An axiomatic approach to the measurement of poverty reduction failure," *Economic Modelling*, Vol. 35, pp. 874–880, DOI: http://dx.doi.org/https://doi.org/10.1016/j.econmod.2012.12.002.
- Corominas-Murtra, Bernat, Joaquín Goñi, Ricard V Solé, and Carlos Rodríguez-Caso (2013) "On the origins of hierarchy in complex networks," *Proc. Natl. Acad. Sci. U. S. A.*, Vol. 110, pp. 13316–13321.
- Cowell, Frank (2016) "Inequality and Poverty Measures," in The Oxford Handbook of Well-Being and Public Policy: Oxford University Press, DOI: http://dx.doi.org/10.1093/ oxfordhb/9780199325818.013.4.
- Czégel, Dániel and Gergely Palla (2015) "Random walk hierarchy measure: What is more hierarchical, a chain, a tree or a star?" *Sci. Rep.*, Vol. 5, p. 17994.
- D'Agostino, Marcello and Valentino Dardanoni (2009) "The measurement of rank mobility," *Journal of Economic Theory*, Vol. 144, pp. 1783–1803, DOI: http://dx.doi.org/https: //doi.org/10.1016/j.jet.2008.11.003.
- Fields, G.S. and E.A. Ok (1999) "The measurement of income mobility: an introduction to the literature," in J. Silber ed. *Handbook of Income Inequality Measurement*: Kluwer Academic Publishers, pp. 557–596.
- Fix, Blair (2017) "Energy and institution size," PLOS ONE, Vol. 12, p. e0171823, DOI: http://dx.doi.org/10.1371/journal.pone.0171823.
 - (2018) "Hierarchy and the power-law income distribution tail," *Journal of Computational Social Science*, Vol. 1, pp. 471–491, DOI: http://dx.doi.org/10.1007/s42001-018-0019-8.
 - (2019) "Personal Income and Hierarchical Power," *Journal of Economic Issues*, Vol. 53, pp. 928–945, DOI: http://dx.doi.org/10.1080/00213624.2019.1657746.
- Graeber, David and David Wengrow (2021) *The Dawn of Everything: A New History of Humanity*: Penguin Press.

- Jäntti, Markus and Stephen P Jenkins (2015) "Income mobility," in *Handbook of Income Distribution*: Elsevier, pp. 807–935.
- Kanbur, Ravi and Diganta Mukherjee (2007) "Poverty, relative to the ability to eradicate it: An index of poverty reduction failure," *Econ. Lett.*, Vol. 97, pp. 52–57.
- Krackhardt, D. (1994) "Graph theoretical dimensions of informal organizations," in *Computational Organization Theory*: Taylor & Francis, pp. 89–111.
- Luo, Jianxi and Christopher L Magee (2011) "Detecting evolving patterns of self-organizing networks by flow hierarchy measurement," *Complexity*, Vol. 16, pp. 53–61.
- Lydall, H. F. (1959) "The Distribution of Employment Incomes," *Econometrica*, Vol. 27, pp. 110–115.
- Maasoumi, E. (1998) "On mobility," in Giles D. and A. Ullah eds. *Handbook of Applied Economic Statistics*: Taylor & Francis, pp. 119–175.
- Mones, Enys, Lilla Vicsek, and Tamás Vicsek (2012) "Hierarchy Measure for Complex Networks," PLOS ONE, Vol. 7, pp. 1–10, DOI: http://dx.doi.org/10.1371/journal. pone.0033799.
- Roberts, David R. (1956) "A General Theory of Executive Compensation Based on Statistically Tested Propositions," *The Quarterly Journal of Economics*, Vol. 70, pp. 270–294.
- Sen, A. (2017) *Collective Choice and Social Welfare*, Cambridge, MA: Harvard University Press.
- Simon, Herbert A. (1957) "The Compensation of Executives," Sociometry, Vol. 20, pp. 32–35.
- Trusina, Ala, Sergei Maslov, Petter Minnhagen, and Kim Sneppen (2004) "Hierarchy measures in complex networks," *Phys. Rev. Lett.*, Vol. 92, p. 178702.
- Turchin, Peter and Sergey Gavrilets (2009) "Evolution of Complex Hierarchical Societies," Social Evolution & History, Vol. 8, pp. 167–198.
- Wright, Jack (2024) "The hierarchy in economics and its implications," *Econ. Philos.*, Vol. 40, pp. 257–278.